

ISSN 2042-2695



No. 1977 January 2024

Does subsidising business advice improve firm performance? Evidence from a large RCT

Gonzalo Nunez-Chaim Henry G. Overman Capucine Riom



THE LONDON SCHOOL OF ECONOMICS AND POLITICAL SCIENCE



Economic and Social Research Council

#### Abstract

We evaluate the impact of the UK's Growth Vouchers Programme, which offered subsidised business advice to 15,207 randomly selected small and medium size enterprises. Using administrative and survey data, we show that the programme increased turnover by 8.2% but only in the short-term and potentially at the expense of non-supported firms. We find that subsidised advice appears to improve firms' capabilities and practices in a way that is consistent with the increase in turnover. We also demonstrate that propensity score matching introduces a sizeable upward bias to estimated effects on turnover and employment and that this bias grows over time.

Keywords: firm performance, enterprise growth, entrepreneurship JEL Codes: D22; D24; L25; L26; M13; O12

This paper was produced as part of the Centre's Growth Programme. The Centre for Economic Performance is financed by the Economic and Social Research Council.

We acknowledge the efforts of many at the (then) Department of Business, Innovation and Skills (BIS) and at its delivery partners who were involved in designing and implementing the Growth Vouchers Programme. We would like to thank colleagues at the Department for Business, Energy and Industrial Strategy (BEIS) and its successor, the Department for Business and Trade (DBT), for comments and help in accessing IDBR and the GVP data, in particular Anastasia Aniwa, Ben Shahlavi, Caroline Booth, Darren Barton, Emily Sumner, Heidi Grainger, Lauren Haycocks, Oliver Wright, Stephanie Walker and Steve Dempsey. We also thank Felicity Algate for comments, as well as James Phipps for his invaluable contributions and multiple conversations that greatly enriched our understanding of the design and implementation of the programme. We are also grateful for the input provided by seminar participants at the University of Oslo and participants to the workshop on Regional and Urban Economics organized by the Center for Regional Economic Development (CRED) and the Department of Economics at the University of Bern. Thanks to Nicolás González-Pampillón for helpful comments and advice, and to Matias Belacín, Carmen Mira-Alonso and Katharina Ziegler for excellent research assistance. The experiment that generated the programme data was registered in the American Economic Association (AEA) RCT Registry under the ID AEARCTR-0000227 and AEARCTR-0008622. This work includes analysis based on data from the IDBR, produced by the Office for National Statistics (ONS). The data is Crown copyright and reproduced with the permission of the controller of HMSO and Queen's Printer for Scotland. The use of the ONS statistical data in this work does not imply the endorsement of the ONS in relation to the interpretation or analysis of the data. This work uses research datasets that may not exactly reproduce National Statistics aggregates. No new data was generated during this study. Supporting data was made available to us by BEIS, subject to a confidentiality agreement. This paper was developed by the authors in their roles at the What Works Centre for Local Economic Growth (What Works Growth), which is funded by the Economic and Social Research Council (ESRC), the Department for Business and Trade, the Department for Levelling Up, Housing and Communities and the Department for Transport. It represents the views of the authors, not What Works Growth or its funders. This work was supported by the ESRC [grant numbers ES/L003945/1, ES/T007702/1]. For the purpose of open access, the authors have applied a Creative Commons Attribution (CC BY) to any Author Accepted Manuscript version arising.

Gonzalo Nunez-Chaim and Capucine Riom, Centre for Economic Performance and What Works Centre for Local Economic Growth both at London School of Economics. Henry G. Overman, London School of Economics, and Centre for Economic Performance and What Works Centre for Local Economic Growth both at London School of Economics.

Published by Centre for Economic Performance London School of Economic and Political Science Houghton Street London WC2A 2AE

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means without the prior permission in writing of the publisher nor be issued to the public or circulated in any form other than that in which it is published.

Requests for permission to reproduce any article or part of the Working Paper should be sent to the editor at the above address.

# 1. Introduction

Productivity in the UK has been stagnant since 2008 and is often characterised by a distribution with a 'long tail' of low-productivity firms (De Loecker, Obermeier & Van Reenen, 2022). As a result, greater attention has been put on improving firms' performance, reflected in high demand for consulting and business support programmes.<sup>1</sup> Public support for business advice can be justified if advice improves firm performance and if market failures, including information asymmetries or externalities, cause firms to under-invest in advice. The UK's Growth Vouchers Programme (GVP) was a large-scale randomised controlled trial set-up to assess the effects on firm performance from tackling three main barriers that firms face when accessing advice: the perception that strategic advice is not needed, uncertainty about where to find good advice and the cost of purchasing advice (BIS, 2014).

To do this, GVP offered diagnostic support and subsided the cost of accessing advice from a directory of advisors, for small businesses in the UK between January 2014 and March 2015. Firms were eligible if they had been trading for at least one year, had less than 50 employees and had not paid for business advice in the previous three years. The programme offered online or personal diagnostic support to all applicants and subsidised the costs of purchasing advice for a group of randomly selected firms. Treated firms were offered a voucher, that would cover half of the cost of advice purchased, up to £2,000. Of 28,158 firms registering online, 20,283 firms completed the application and 75% of these firms were randomised to treatment, 25% to a control group.

The large sample size and experimental set up allows us to estimate the causal effect of receiving and using a voucher and the protocol envisaged exploring heterogeneity by type of intervention.<sup>2</sup> We assess impacts using administrative data from the Inter-Departmental Business Register (IDBR), the most comprehensive source of data on firms in the UK, covering around 2.7 million firms.<sup>3</sup> We complement this with information from baseline and endline survey data collected 12 and 24 months after firms received diagnostic support.

<sup>&</sup>lt;sup>1</sup> Consulting services are expected to grow around 13% in 2022 according to a survey from the Management Consultancies Association (https://www.mca.org.uk/press-releases/growth-for-uk-consulting). Around 40% of SMEs in the UK used some type of formal business assistance between 2008 and 2011, and nearly half of them relied on a publicly funded programme (North et al., 2011).

 $<sup>{}^2\</sup> https://www.gov.uk/government/publications/growth-vouchers-programme-trial-protocol.$ 

<sup>&</sup>lt;sup>3</sup> The IDBR data available to us corresponds to the first quarter of every year between 2012 and 2019.

The IDBR allows us to consider the impact of the programme on turnover and employment. Firms that used their voucher saw an 8.2% increase in turnover after 12months, but this effect did not persist at 24-months. The programme had no effect on employment. We find similar results using survey data on firm performance collected by GVP. The survey also provides evidence on capabilities and practices adopted, and we argue that these are consistent with the finding of a positive effect on turnover, mainly due to the development of new products, rather than improved access to the UK or other markets. One concern is the extent to which the effect on turnover represents displacement since between 69% and 76% of firms that used the voucher are likely to serve local markets.<sup>4</sup>

In line with the trial protocol, we explore heterogeneity by type of diagnostic received and theme of advice selected. Caution is needed here given the diagnostic was a small component of the programme and the theme was selected by the firm. We find no significant differences in effects on turnover and employment between types of diagnostic. For theme of advice, the only significant results is among firms that received business advice on sales & marketing, whose turnover increased by 18.8% and 19.8% after 12 and 24 months, respectively. Results are inconclusive when undertaking heterogeneity analysis by firm size and age.

We contribute to the literature using randomised controlled trials (RCTs) to assess the impact of business support. Much of the existing evidence involves small trials in developing countries focused on the impact of providing training and consulting. Our research fills an important gap by looking at subsidising business advice in a developed country in a large trial (15,207 randomly selected firms). Overall, the literature suggests there is heterogeneity in the effectiveness of training and consulting projects. Moreover, even though some studies find positive effects on practices adopted, managerial capabilities and other intermediate outcomes, such as increased business knowledge, the impact on profit or revenue often turns out to be statistically insignificant (Mano et al., 2012; Valdivia, 2015; Bruhn & Zia, 2013; Karlan & Valdivia, 2011; Karlan, Knight and Udry, 2015; Bardasi et al., 2021). McKenzie and Woodruff (2014) provided a review and

 $<sup>^4</sup>$  We assign firms to tradable and non-tradable sectors based on their SIC 4 digit code (see appendix section A.5). From the group of firms that received and used the voucher, 46.4% are in medium-tradable, 23.0% in non-tradable services, and 6.6% in other sectors, which would be considered as non-tradable by some definitions.

highlight several key issues including small sample sizes with a group of heterogeneous firms and a focus on short time horizons (usually outcomes within a year).

The number of experiments in developed countries has so far been limited. Benus et al., (1994) evaluate two RCTs in the United States offering entrepreneurial training, business support and financial assistance. They find that these programmes increased the likelihood of establishing a new business after nearly 2 years. Also in the United States, project Growing America through Entrepreneurship (GATE) was implemented as a randomised controlled trial providing an assessment of business needs, classroom training, counselling, and assistance in applying for financing to 4,197 individuals. GATE's official project evaluation from Benus et al. (2010) finds that participants were more likely to start a business, although businesses started by the treated group did not experience higher growth and profitability. Fairlie, Karlan and Zinman (2015) further analyse project GATE and find no effect on business sales, earnings, employment, or firm survival in the long-term. Contrary to this, Georgiadis and Pitelis (2016) report positive effects on firm performance of randomly assigned training for employees and managers in a sample of 1,325 small and medium-sized enterprises in the UK. A small number of RCTs in developed countries look at training for start-ups. See, for example Nagel, Huber, Van Praag & Goslinga (2019) using mandatory tax training in the Netherlands provided to a randomly selected group of first-time entrepreneurs or Camuffo et al. (2020) testing the effect of teaching a scientific approach to business performance on 116 Italian startups.

We also contribute to the literature by comparing our estimates from the randomised controlled trial to those obtained from propensity score matching (PSM) and estimate the magnitude of the selection bias in this context. To the best of our knowledge, this is the first study that provides additional evidence on the magnitude of the bias introduced by PSM in the context of firm-level interventions. LaLonde (1986) was one of the first to undertake a similar exercise for National Supported Work (NSW), a labour training programme. Dehejia and Wahba (1999, 2002) build on LaLonde (1986) using the kind of PSM techniques we use below. See also Smith and Todd (2001, 2005). Heckman et al. (1996, and 1998) and Heckman et al. (1997) do something similar for another training programme (the US National Job Training Partnership Act). Michalopoulos, Bloom and Hill (2004) compare experimental estimates with PSM results from welfare-to-work programmes in the United States. Such exercises are increasingly common in the education literature. For example, Agodini and Dynarski (2004) compare PSM and

experimental estimates for the School Dropout Demonstration Assistance programme in the United States; Wilde and Hollister (2007) compare experimental results from the Tennessee's Student Teacher Achievement Ratio Project (STAR Project) with PSM estimates. Diaz and Handa (2006) conclude that PSM results are only similar to experimental estimates for some outcomes. Weidmann and Miratrix (2021) examine 14 studies for a set of small school-level interventions across primary schools in England. They find no evidence of substantial selection bias due to unobserved characteristics, in part due to a large overlap between treatment and control groups in terms of observed covariates.

Our research provides additional evidence on the magnitude of the bias introduced by PSM in the context of firm-level interventions. Even when using comprehensive administrative data and achieving good matching properties, our PSM results suggest positive and persistent impacts of the programme on both turnover and employment in every period, contrary to the results from exploiting the RCT. However, if we exclude the selection effect and isolate the main aspect of the programme, the subsidised advice component, the results indicate small and temporary effects, consistent with our experimental estimates. This suggests that the bias introduced from unobservables is not only large, but increases over time. Overall, these results highlight the importance of evaluating the impact of business support using a robust methodology, ideally a randomised controlled trial.

The remainder of this paper is structured as follows: In the next section, we describe the Growth Vouchers Programme. Section 3 presents the experimental set up and identification strategy, while Section 4 discusses data sources. Section 5 contains our results, examining the impact of the programme on firm performance, exploring channels through which firms improved their outcomes, and looking at heterogeneous treatment effects. Section 6 compares our experimental results to those obtained from propensity score matching approach and provides evidence on the bias introduced when not considering unobservable characteristics. Finally, Section 7 concludes.

# 2. The Growth Vouchers Programme

The UK's Growth Vouchers Programme (GVP) was designed and implemented by the Department for Business, Innovation and Skills (BIS) and the Behavioural Insights Team

(BIT) in 2014 and 2015.<sup>5</sup> GVP was set up as a randomised controlled trial and aimed to encourage firms to take-up business advice and to help them grow.<sup>6</sup> Firms were eligible if they had been trading for at least one year, had less than 50 employees and had not paid for advice in the previous three years.<sup>7</sup> BIS and its delivery partners promoted the programme using a range of marketing activities, including email campaigns, telemarketing and social media.<sup>8</sup>

The programme offered online or personal diagnostic support to all applicants and subsidised the costs of purchasing advice for a group of randomly selected firms. Randomisation occurred at two points and determined whether the firm was offered an online or personal diagnostic and whether the firm received a voucher to subsidise advice. The next section describes this randomisation in detail. The remainder of this section describes each step of the programme.

Both diagnostics aimed to understand the needs and preferences of firms for receiving advice in one of five areas, rather than providing a comprehensive diagnostic and imposing a recommended theme of advice.<sup>9</sup> GVP was structured around five areas or 'themes' – sales & marketing, raising finance, leadership & management, workforce development and digital technologies.<sup>10</sup> The online diagnostic involved a short questionnaire covering the five themes of advice. For the personal diagnostic, firms were allocated to regional delivery partners who had to offer a diagnostic shortly after application.<sup>11</sup> Firms were offered the option of undertaking the diagnostic face-to-face, or via telephone or Skype.<sup>12</sup>

The personal diagnostic involved a greater time commitment than the online diagnostic (which took around 10-15 minutes). This may explain why completion rates differed by the type of diagnostic -62% for personal, 83% for online. Firms receiving the personal

<sup>&</sup>lt;sup>5</sup> BIS is now the Department for Business, Energy and Industrial Strategy (BEIS).

<sup>&</sup>lt;sup>6</sup> See page 4 of the trial protocol https://www.gov.uk/government/publications/growth-vouchers-programme-trial-protocol.

<sup>&</sup>lt;sup>7</sup> Self-reported eligibility was checked by delivery partners. See page 9 of the trial protocol.

<sup>&</sup>lt;sup>8</sup> BIS commissioned delivery partners across the UK to recruit firms onto the programme through marketing activities, check eligibility of applicants, carry out diagnostic assessments and assess claims for funding.

<sup>&</sup>lt;sup>9</sup> BIS and BIT were concerned about the depth of the firm diagnostics, and worried that imposing a theme of advice recommended by delivery partners could discourage voucher take-up.

 $<sup>^{10}</sup>$  This is the ordering of themes used in the online questionnaire.

<sup>&</sup>lt;sup>11</sup> Delivery partners had to carry out the diagnostics within two weeks of a firm applying, but this rule was later relaxed. BIS gave delivery partners a broad remit in the design of personal diagnostic assessments.

<sup>&</sup>lt;sup>12</sup> From the 9,488 firms allocated to a personal diagnostic, 42% received it face-to-face, while 58% over the phone or via Skype. Differences in resources and capacity meant variation across delivery partners in the style, length, and content of assessments. Most firms receiving a personal diagnostic reported that the conversation with the advisor helped clarify areas for improvement (BIS, 2014).

diagnostic discussed their needs and preferred theme of advice with the delivery partner. For the online diagnostic, the theme of advice was presented to the firm upon completion of the self-assessed online diagnostic, based on their responses to questions in each area.

Firms then selected a theme of advice, and some were randomly offered a voucher, capped at £2,000, that would cover up to 50% of the cost of purchasing advice. All firms were then referred to an online directory of advisors – developed and run by Enterprise Nation, a business support network. Although voucher randomisation was conditional on the theme of advice suggested by the diagnostic, firms were not restricted to accessing that kind of advice. Firms did not have to use the voucher to pay for the service. Those that were not allocated a voucher had to pay the full cost of any advice received.

The online directory contained details of thousands of consultants in each of the five areas of advice, listed by the type of support they offered. Consultants were selected for the directory based on their experience, qualifications and membership of professional bodies (BIS, 2014). They provided bespoke advice which generally involved assistance over weeks or months. The consultant usually reviewed information about the structure and performance of the firm, before discussing improvements, targets and how to monitor progress. The intensity of the support varied and often involved working closely with the consultant to develop a business plan or a marketing strategy or explore financing alternatives.

# 3. Experimental set up and identification

The trial protocol allowed for firms to be randomised at two points, determining the type of diagnostic and whether the firm received a voucher (see Figure 1). The protocol determined that 25% of eligible firms were to be allocated to the online diagnostic and 75% to the personal diagnostic. The diagnostic assessment resulted in firms choosing a theme of advice. Within each theme, firms were to be randomly allocated to a treatment or control group with 75% of firms allocated to the treatment group and 25% to the control group.<sup>13</sup>

The treatment group received a voucher that could be used to cover 50% of the cost of business advice purchased from an online directory of consultants. Although firms could

<sup>&</sup>lt;sup>13</sup> A stratified randomised design by business sector was not feasible due to the lack of accurate information by sector (BIS, 2014).

spend as much on advice as they wanted, the value of the voucher was capped at £2,000. Overall, the mean value claimed was £1,714, against an average total purchase of £3,713 for advice.<sup>14</sup> The control group only received the diagnostic (online or personal, depending on the initial randomisation) and a referral to the same online directory.



Figure 1. Timeline of the programme

Note: The diagram shows the journey of firms from application to the allocation of vouchers. Source: Based on GVP's full trial protocol, available at https://www.gov.uk/government/publications/growth-vouchers-programme-trial-protocol.

As with any voluntary programme, unobservables might drive self-selection in terms of who applies and completes the application, which has some implications for external validity. However, if the programme had been implemented as planned, then conditional on completing the application, the two-step randomisation would have allowed us to estimate the effect of type of diagnostic and of subsidising the cost of purchasing business advice (via a voucher) between treated and control firms. In practice, several complications arise due to changes in implementation and issues with compliance.

First, the random allocation of firms to each type of diagnostic broke down in the last month (March 2015) when many firms applied in response to a large marketing campaign, initiated because of concerns over enrolment. Around 31% of the total number of applications were submitted in this month, surpassing the capacity to deliver personal diagnostics. As a result, the allocation ratio between personal and online diagnostics was not followed, and a larger share of applicants received an online diagnostic.

<sup>&</sup>lt;sup>14</sup> One third of firms that received and used the voucher claimed less than  $\pounds 2,000$ . For this group, the mean total cost of advice received was  $\pounds 2,489$ . The remaining 67% of firms claimed the maximum value of  $\pounds 2,000$  and paid an average of  $\pounds 4,311$  in total.

Second, around 33% of those receiving an online diagnostic picked a theme of advice different to that suggested by the outcome of the online questionnaire. While the headline statistic suggest self-selection was only an issue for the online diagnostic, it may also be an issue for the personal diagnostic as the theme was selected by the firm in discussion with the advisor. The potential self-selection into a type of advice limits our ability to draw conclusions on the causal impact of GVP by theme of advice. While the protocol was not followed in terms of the random allocation to diagnostic, random allocation of the vouchers, conditional on selected theme of advice, was implemented as planned. This is illustrated in Figure 2 which presents the breakdown of firms that selected each theme, as well as the allocation to treatment and control within each of them.

A third issue with the implementation of the programme was the low take up of the GVP voucher. Only 33% of firms that were randomly allocated a voucher used it.<sup>15</sup> Comparing voucher recipients to non-recipients provides an estimate of the intention-to-treat effects (see Figure 3 for a description of the different groups of firms). The low take-up means that this may not be particularly informative in terms of the impact of subsidised advice on firm performance. For this reason, we also estimate the effect of treatment on the treated using an instrumental variables (IV) approach which exploits the variation in take up of the voucher induced by the random allocation of the voucher.





Note: The grey bars show the distribution of firms among the five themes of advice. The blue bars depict the allocation of firms to treatment and control group within each theme of advice. Source: Author calculation using baseline information.

To account for the breakdown of the random assignment to diagnostic, our preferred specifications include the full set of interactions between i) the type of diagnostic received;

<sup>&</sup>lt;sup>15</sup> Firms that used the voucher were more likely to be able to afford the cost of business advice, since at baseline they have a higher turnover and more employees compared to non-compliers.

ii) the theme of advice selected; and iii) a dummy indicating if the firm applied in the last month. We also include a set of basic controls which correspond to month of application and cohort of firms to improve precession by accounting for potential differences in diagnostic and treatment delivery across cohorts and over time. For more details on the identification strategies, see appendix section A.4.



Figure 3. Group of applicants, recipients and compliers

Note: The diagram shows the split of firms depending on whether they were eligible, applied to the programme and used the voucher. Source: Author calculation using baseline information.

# 4. Data

Our analysis uses secondary data from the Inter-Departmental Business Register (IDBR) and survey data collected by the GVP programme. IDBR provides data on employment and turnover and can be used to calculate survival rates. IDBR is a live register held by HM Revenue and Customs and constructed via Pay As You Earn (PAYE) and Value Added Tax (VAT) records. It covers all firms except those with no employees or that are below the VAT threshold. Firms on the IDBR account for almost 99 percent of UK economic activity (ONS, 2019).

The version provided to us by the Department for Business, Energy and Industrial Strategy (BEIS) represents a point in time snapshot for the first quarter of every year from 2012 to 2019. Our analysis considers the sample of applicants from the GVP programme that we can identify in the IDBR – we manage to match around 67% of firms.<sup>16</sup> Secondary data has a reporting lag of 1 year for employment, and 2 years for turnover; thus, we can only estimate effects for 2016-2018 on employment, 2016-2017 on turnover. Firm outcomes are not updated for every firm in every year, so we use the group of firms with updated data on each outcome in each year.<sup>17</sup>

<sup>&</sup>lt;sup>16</sup> The unique code of applicant firms provided to us does not allow us to identify some of them in the IDBR. <sup>17</sup> This does not alter the balance of firms allocated to treatment and control groups. For all outcomes and periods, the proportion of treated firms ranges between 74.4% for those with updated data on employment from the 12-month IDBR and 75.0% among firms with updated data on turnover from the 12-month IDBR.

That is, we do not impose a sample restriction to consider a balanced panel. The data appendix (section A.1) has further details on variable definitions, cleaning of the data and how we deal with the small number of firms that appear in the GVP programme data even though the IDBR data suggest they are ineligible.

Survey data is available at baseline, and 12 and 24 months after diagnostic. Follow upinterviews of treated and control firms covering firm performance, expectations, capabilities, and practices were done 12 and 24-months after firms had completed the diagnostic.<sup>18</sup> Performance was self-reported by firms and includes turnover, number of employees and establishments, among other outcomes. The sampling strategy followed a census approach for the baseline and 12-month survey data, while the 24-month survey contacted every participant who responded to the 12-month survey. Baseline information was collected for 28,158 firms via an online questionnaire during the application process. Of these, 20,283 completed the diagnostic and application process and 7,743 did not. 5,623 firms completed the 12-month survey.<sup>19</sup> From these, 1,982 firms completed the 24-month survey (the rest either were not reached or did not complete the survey). Since the application period ran from January 2014 to March 2015, interviews were conducted in two cohorts to collect information approximately 12 and 24 months after receiving the diagnostic, regardless of the month of application. Remember, that all firms receive a diagnostic, while only some firms receive a voucher to cover for business advice and that the latter is the treatment of interest.

Gaps between the date of diagnostic and receiving the advice, and the timing of both relative to IDBR and survey data introduce variation in timing of the outcome data relative to treatment, as summarised in Table 1.<sup>20</sup> For example, the 12-month survey reports data that, on average, comes only 7.9 months after the voucher was claimed, while the gap is 9.7 months for IDBR. In what follows we discuss results at 12, 24 and 36 months

<sup>&</sup>lt;sup>18</sup> Eight capabilities covered: managing people; developing a business plan; accessing the UK market; accessing other markets; developing new products; accessing external finance; recruiting new employees; using IT to grow. Nine practices covered having a: business plan; marketing plan; marketing budget; corporate website; workforce strategy; recruitment budget; training budget; cash flow forecast; regular financial review.

<sup>&</sup>lt;sup>19</sup> A consulting firm conducted the telephone surveys using a Computer Assisted Telephone Interviewing system. The questionnaires included skip patterns, dynamic checks and constraints to ensure the quality of information collected. Interviewers were randomly allocated to businesses and were supervised during data collection.

<sup>&</sup>lt;sup>20</sup> 42 firms were allocated and used the voucher, but at the time of the 12-month survey had not received the business support. Hence, these firms are considered as non-participants in estimates after 12 months and as participants after 24 months. Similarly, 3 participants claimed the voucher after the 1st quarter of 2016, which corresponds to the date of IDBR 2016. So, they are also considered as non-participants in estimates after 12 months and as after 12 months and as participants in subsequent periods.

while acknowledging that this is a simplification and abstracts from variation seen in both surveys. More details on how we consider the timing of the treatment in relation to the data are in the data appendix (A.1).

Table 1. Thing of ou	teome aata i		ini date or	the voucher				
Number of months between claim date of the voucher and date of the data used for analysis								
Dataset	Mean	Median	s.d.	Min.	Max.			
12-month survey	7.9	7.7	2.3	3.1	14.0			
24-month survey	19.6	19.4	2.4	3.8	25.8			
12-month IDBR	9.7	8.9	3.2	0.3	20.5			
24-month IDBR	21.7	20.9	3.2	12.3	32.5			
36-month IDBR	33.7	32.9	3.2	24.3	44.5			

Table 1. Timing of outcome data relative to claim date of the voucher

Note: For the 12/24-months survey the estimates use the mid-point of the period when interviews were conducted for each cohort of firms. For secondary sources, the summary statistics are calculated for firms that claimed the voucher using the mid-point of the period from the IDBR.

Given these issues with timing for the analysis of survey outcomes, we include dummies that capture the month of application and the survey-cohort of firms as our set of basic control variables. Note that, similarly to the analysis of administrative data, we also include the full set of interactions between the type of diagnostic received; the theme of advice selected; and a dummy indicating if the firm applied in the last month to account for the breakdown of the random assignment to diagnostic.

# 5. Results

In 2015, the average firm that completed the GVP application process had been operating for over 9 years, employed 7.3 people and had an annual turnover of around £773,000 (Table A.1). For this group of firms, the median growth in turnover a year prior to the programme (between 2014 and 2015) was 7.4%. Table A.2 to Table A.4 report balancing checks between treatment and control groups using GVP baseline data for the three samples for which we have IDBR, 12 and 24-month survey data, respectively. Each row reports results from a regression of the baseline characteristic on a dummy variable indicating treatment status, our basic set of controls and the full set of interactions, as detailed above.

For the sample of firms that are matched to the IDBR, there are statistically significant differences between treatment and control for three out of 47 variables in the baseline data. There is one statistically significant difference at baseline for the sample of firms in

the 12 month-survey and four for the sample in the 24-month survey. Eight significant differences from a total of 141 balancing tests is in line with what would be expected to find by chance at 5% levels of significance and the magnitude of these differences is relatively small. Controlling for baseline characteristics does not change our results.

#### 5.1 Effect of the programme on firm performance

The IDBR provides the most reliable source of data for firm performance – it is subject to more cross-checking than the self-reported survey data and although the matching rate from GVP to IDBR is around 67%, there is no reason to think that this is related to participation in GVP. Table 2 reports estimates from ordinary least squares (OLS) and IV turnover and employment from the 12, 24 and 36-month IDBR.<sup>21</sup> First-stage results for the IV are in Table A.5. The coefficient for voucher allocation is positive and statistically significant in all the samples considered.<sup>22</sup> As expected, the first-stage estimates indicate that the allocation of vouchers induced firms to use the voucher given to them.

For both OLS and IV, column 1 in Table 2 presents results from a specification with no controls. Column 2 controls for type of diagnostic and theme of advice as well as for our basic set of controls, while column 3, our preferred specification, includes all the interactions between type of diagnostic, theme of advice and the last month of application, and the basic set of controls.<sup>23</sup> All estimates on turnover and employment are conditional on firm survival.<sup>24</sup> As discussed above the OLS results provide estimates of the intention-to-treat (ITT) and the IV results provide estimates of the average treatment effect on the treated (ATT).

Effects on employment are negative, small in magnitude and insignificant in all periods. For both outcomes, the IV coefficient is about three times as large as the OLS coefficient after 12 and 24 months, but they are both zero in the 36-month IDBR. Despite

 $<sup>^{21}</sup>$  As discussed in section 3, the 12-month IDBR reports outcomes somewhere between 9.7 and 13.1 months after treatment. Similarly, on average the 24-month IDBR reports outcomes between 21.7 and 25.1 months after treatment, and the 36-month IDBR reports outcomes between 33.7 and 37.1.

<sup>&</sup>lt;sup>22</sup> A larger coefficient in the first stage from survey data relative to the IDBR reflects a larger proportion of compliers captured in the former sample compared to the latter. In the same way, the 24-month survey sample includes a larger share of compliers relative to the 12-month survey. The split of firms in survey data is 76% treated and 24% control in the 12-month survey, and 80% treated and 20% control in the 24-month survey.

<sup>&</sup>lt;sup>23</sup> From the overall group of participants, we can identify 13,444 firms in the IDBR. However, only 9,452 firms have updated data on turnover from the 12-month IDBR and 8,855 from the 24-month IDBR. Similarly, for employment only 7,326 firms have updated data from the 12-month IDBR; 6,606 from the 24-month IDBR; and 6,084 from the 36-month IDBR.

<sup>&</sup>lt;sup>24</sup> While survival is not an outcome considered in the trial protocol, we present ITT and IV estimates for survival in Table A.6 as turnover and employment data is conditional on firm survival. The results suggest GVP had no impact on the survival of firms in any period.

voucher take-up of only 33%, the OLS estimate from receiving a voucher indicates a positive effect on turnover of 3% after 12 months. The IV estimate for firms that received and used the voucher is 8.2%. This corresponds to an increase of £73,120, given the average turnover of compliers in 2015. The positive effect on turnover is short-lived – the OLS and IV coefficients at 24 months are smaller (0.9% and 2.4% respectively) and statistically insignificant.

	OLS	OLS	OLS	IV	IV	IV
	(1)	(2)	(3)	(1)	(2)	(3)
		T	urnover aft	er 12 mont	hs	-
Allocation of voucher	0.030* (0.014)	0.030* (0.014)	0.030* (0.014)	N/A	N/A	N/A
Take up of the voucher	N/A	N/A	N/A	0.083* (0.039)	$0.082^{*}$	$0.082^{*}$
Obs.	9.452	9.452	9.452	9.452	9.452	9.452
	- , -	<i>T</i>	urnover aft	er 24 mont	hs	- / -
Allocation of voucher	0.010 (0.022)	0.009 (0.022)	0.009 (0.022)	N/A	N/A	N/A
Take up of the voucher	N/A	N/A	N/A	0.027 (0.061)	0.025 (0.061)	0.024 (0.061)
Obs.	8,855	8,855	8,855	8,855	8,855	8,855
		Em	ployment a	fter 12 moi	nths	
Allocation of voucher	-0.008 (0.007)	-0.009 (0.007)	-0.009 (0.007)	N/A	N/A	N/A
Take up of the voucher	N/A	N/A	N/A	-0.025 (0.020)	-0.025 (0.020)	-0.026 (0.020)
Obs.	7,326	7,326	7,326	7,326	7,326	7,326
		Em	ployment a	fter 24 mol	nths	
Allocation of voucher	-0.007 (0.011)	-0.007 (0.011)	-0.008 (0.011)	N/A	N/A	N/A
Take up of the voucher	N/A	N/A	N/A	-0.021 (0.032)	-0.021 (0.031)	-0.022 (0.032)
Obs.	6,606	6,606	6,606	6,606	6,606	6,606
	· · ·	Em	ployment a	fter 36 mor	nths	
Allocation of voucher	0.000 (0.015)	0.001 (0.015)	-0.000 (0.015)	N/A	N/A	N/A
Take up of the voucher	N/A	N/A	N/A	0.000 (0.043)	0.002 (0.043)	-0.000 (0.043)
Obs.	6,084	6,084	6,084	6,084	6,084	6,084
Basic controls	No	Yes	Yes	No	Yes	Yes
Type of diagnostic & theme of advice	No	Yes	N/A	No	Yes	N/A
Full set of interactions	No	No	Yes	No	No	Yes

Table 2. OLS and IV estimates for firms in the IDBR

Note: OLS and IV regressions for percentage change since baseline. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05. Basic controls include month of application and cohort. The full set of interactions corresponds to all combinations between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. The result on turnover after 12 months is significant at a 7% level when applying the Romano-Wolf multiple hypothesis correction for turnover and employment with 1,000 bootstrap repetitions. N/A: not applicable.

The OLS and IV estimates for turnover and employment are robust to several sensitivity checks. First, we restrict the analysis to the group of firms with updated data in all three periods, thus considering the same group of firms after 12, 24 and 36 months

(Table A.7). The positive IV effect on turnover after 12 months is slightly smaller (7.1%) and significance drops (to around 6%). The effects for turnover at 24 months and for employment are also unchanged. Second, we exclude firms that applied in the last month when the random allocation between type of diagnostic was not followed (Table A.8). The results are similar across all variables and time periods – the IV effect on turnover is 10.6% after 12 months. Third, we exclude firms that did not select the recommended theme of advice (Table A.9). We also find similar effects for all variables and periods, including a positive IV coefficient for turnover of 10.2% after 12 months. Finally, we apply the Romano-Wolf multiple hypothesis correction (Romano and Wolf, 2005a & 2005b) using a step-down procedure for our two outcomes with 1,000 bootstrap repetitions to derive standard errors. Our IV result on turnover after 12 months remains significant, albeit at a 7% level.

The 12-month survey identifies 99 always-takers – firms that were not allocated a voucher but purchased business advice from the GVP online directory. We consider these control firms as participants in IV estimates in all the periods as an additional robustness check (Table A.10).<sup>25</sup> The IV effect is now 8.9% on turnover after 12 months, with all other results continuing to be insignificant. Finally, the survey data identifies firms that received any other business advice paid at a commercial rate, or free or subsidised advice as part of other government schemes. These firms received support similar to GVP, but are not always-takers since that is defined in relation to the allocation of a GVP voucher. Thus, we expand the definition of participation into the programme by considering all firms that either i) paid for advice from the GVP online directory; or ii) received any other type of advice (Table A.11).<sup>26</sup> Using this definition, the IV estimate is significant and around the same magnitude (9.1%) for turnover after 12 months, with all other results continuing to be insignificant.

#### 5.2 Self-reported outcomes, expectations, capabilities, and practices

Estimates for self-reported outcomes – the number of establishments, choice to begin exporting, increase in turnover, employment and growth expectations – from the 12- and

<sup>&</sup>lt;sup>25</sup> The 24-month questionnaire did not include the question on whether firms had purchased advice from the GVP online directory or not. Thus, we cannot identify control firms that paid for GVP advice from the 24-month survey.

<sup>&</sup>lt;sup>26</sup> The survey data allows us to identify firms that received any other type of business advice. Those captured from the 12-month survey are considered as participants in the 12, 24 and 36-month IDBR estimates, while those identified from the 24-month survey are considered as participants only in the 24 and 36-IDBR estimates.

24-month surveys relative to baseline provide supporting evidence for a short-term effect on turnover relative to baseline (Table 3 and Table 4).<sup>27</sup> There is no effect on number of establishments, employment or beginning to export. Effects on turnover are consistent with IDBR – positive and significant at 12-months, smaller and insignificant at 24 months.<sup>28</sup> Note that the coefficients are not directly comparable to those from IDBR as the survey asks whether turnover increased, decreased or stayed the same, while IDBR reports values. The effect on growth expectations is positive in both periods, although we do not know whether these changes in sentiment are a direct effect of the advice, or an indirect effect resulting from the increase in turnover at 12-months.

	OLS	IV	OLS	IV
	(1)	(2)	(3)	(4)
	Number of es	stablishments	Begin e.	xporting
Allocation of voucher	-0.009 (0.007)	N/A	0.003 (0.011)	N/A
Take up of the voucher	N/A	-0.020 (0.015)	N/A	0.008 (0.024)
Obs.	3,614	3,614	5,579	5,579
	Increase i	n turnover	Employment	
Allocation of voucher	0.040** (0.016)	N/A	-0.125 (0.113)	N/A
Take up of the voucher	N/A	0.092** (0.036)	N/A	-0.257 (0.231)
Obs.	5,499	5,499	3,608	3,6,08
	Growth ex	spectations		
Allocation of voucher	0.101* (0.040)	N/A		
Take up of the voucher	N/A	0.206* (0.083)		
Obs.	3,573	3,573		

Table 3. OLS and IV estimates on self-reported outcomes for firms in the 12-month survey

Note: OLS and IV estimates from our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

Table 5 and Table 6 report results for a range of business capabilities and practices from the 12<sup>-</sup> and 24<sup>-</sup> month surveys. While one could consider these independently of the results on firm performance, we focus on the extent to which they are consistent with the impacts on turnover identified in Table 2 to Table 4. Capabilities three to five – which include accessing UK and other markets and developing new products – appear the most

 $<sup>^{27}</sup>$  As discussed in section 3, on average these survey results are for data a couple of months before the corresponding IDBR data.

<sup>&</sup>lt;sup>28</sup> We do not consider a sample restriction for the sub-set of firms that are included in both IDRB and survey data as this results in a small sample size for both IDBR and survey estimates.

directly relevant for explaining the increase in turnover, so it is reassuring that the only significant increase in capabilities is in this sub-group. The results suggest that development of new products, rather than improved access to the UK or other markets could explain the increase in turnover after 12 months.

	OLS	IV	OLS	IV	
	(1)	(2)	(3)	(4)	
	Number of es	stablishments	Begin ez	xporting	
Allocation of voucher	-0.006 (0.032)	N/A	-0.011 (0.024)	N/A	
Take up of the voucher	N/A	-0.011 (0.063)	N/A	-0.022 (0.046)	
Obs.	1,959	1,959	1,965	1,965	
	Increase i	n turnover	Employment		
Allocation of voucher	0.019 (0.028)	N/A	-0.237 (0.251)	N/A	
Take up of the voucher	N/A	0.037 (0.054)	N/A	-0.464 (0.493)	
Obs.	1,944	1,944	1,953	1,953	
	Growth ex	spectations			
Allocation of voucher	0.144** (0.057)	N/A			
Take up of the voucher	N/A	0.282** (0.112)			
Obs.	1,930	1,930			

Table 4. OLS and IV estimates on self-reported outcomes for firms in the 24-month survey

Note: OLS and IV estimates from our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

For practices, the strongest effect is for improvements in the share of theme-specific practices adopted. Although this is not surprising – the theme of advice and the business advice purchased were selected by firms and tailored to their needs – it is consistent with the effects on turnover as sales & marketing was the most popular theme of advice. Similarly, the only significant effect on specific practices is for having a marketing plan – again consistent with the effect on turnover.

The impact on capabilities remains the same after 24 months: improvements in developing new products, as well as a positive and statistically significant effect on themespecific practices adopted. Firms receiving and using vouchers also move on from having a marketing plan (after 12 months) to having a marketing budget (after 24 months). The results suggest firms tended to maintain their improvements in terms of capabilities and practices adopted after 24 months, but that does not appear to translate in to sustained turnover growth. In summary, improvements in capabilities and adoption of practices could help firms with a one-off increase in turnover to a new higher level (after 12 months), but maintaining these changes is not enough for sustaining growth, which would be reflected as an increase in turnover to an even higher level after 24 months.

	OLS	IV	OLS	IV	OLS	IV
	(1)	(2)	(3)	(4)	(5)	(6)
	Capability .	1: Managing	Capability 2	Developing a	Capability 3:	Accessing the
	ped	ple	busine	ess plan	UK n	narket
Allocation of voucher	0.043 (0.031)	N/A	0.040 (0.031)	N/A	0.006 (0.034)	N/A
Take up of the voucher	N/A	0.087 (0.062)	N/A	0.081 (0.063)	N/A	0.013 (0.070)
Obs.	3,353	3,353	3,562	3,562	2,945	2,945
	Capability other n	4: Accessing narkets	Capability & new p	5: Developing roducts	Capability externa	6: Accessing l finance
Allocation of voucher	0.017 (0.043)	N/A	0.070* (0.033)	N/A	0.024 (0.038)	N/A
Take up of the voucher	N/A	0.037 (0.094)	N/A	0.142* (0.067)	N/A	0.049 (0.080)
Obs.	1,853	1,853	3,285	3,285	2,442	2,442
	Capability 7 new em	7: Recruiting ployees	Capability to g	8: Using IT grow	Share of the practices	eme-specific s adopted
Allocation of voucher	0.000 (0.034)	N/A	0.030 (0.031)	N/A	0.378** (0.064)	N/A
Take up of the voucher	N/A	0.000 (0.069)	N/A	0.061 (0.064)	N/A	0.871** (0.147)
Obs.	3,085	3,085	3,528	3,528	5,579	5,579
	Practice 1 busine	: Having a ss plan	Practice 2: Having a marketing plan		Practice 3 marketii	: Having a ng budget
Allocation of voucher	0.003 (0.019)	N/A	0.047* (0.020)	N/A	0.011 (0.020)	N/A
Take up of the voucher	N/A	0.007 (0.040)	N/A	0.096* (0.041)	N/A	0.023 (0.042)
Obs.	3,615	3,615	3,615	3,615	3,615	3,615
	Practice 4: Having a corporate website		Practice & workforc	5: Having a e strategy	Practice 6 recruitme	Having a ent budget
Allocation of voucher	0.014 (0.012)	N/A	0.024 (0.018)	N/A	-0.014 (0.013)	N/A
Take up of the voucher	N/A	0.029 (0.025)	N/A	0.048 (0.037)	N/A	-0.029 (0.027)
Obs.	3,615	3,615	3,615	3,615	3,615	3,615
	Practice 7	: Having a	Practice 8: 1	Having a cash	Practice 9	): Having a
	training	r budget	flow f	orecast	regular fina	encial review
Allocation of voucher	0.007 (0.019)	N/A	-0.004 (0.018)	N/A	0.018 (0.015)	N/A
Take up of the voucher	N/A	0.015 (0.039)	N/A	-0.009 (0.038)	N/A	0.036 (0.031)
Obs.	3,615	3,615	3,615	3,615	$3,\!615$	3,615

Table 5. OLS and IV estimates on capabilities and practices for firms in the 12-month survey

Note: OLS and IV estimates from our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

	OLS	IV	OLS	IV	OLS	IV
	(1)	(2)	(3)	(4)	(5)	(6)
	Capability .	1: Managing	Capability 2	Developing a	Capability 3:	Accessing the
	ped	ople	busine	ess plan	UK n	narket
Allocation of voucher	-0.001 (0.043)	N/A	0.026 (0.043)	N/A	0.031 (0.048)	N/A
Take up of the voucher	N/A	-0.002 (0.084)	N/A	0.051 (0.084)	N/A	0.061 (0.096)
Obs.	1,813	1,813	1,938	1,938	1,608	1,608
	Capability	4: Accessing	Capability &	5: Developing	Capability	6: Accessing
	other n	narkets	new p.	roducts	externa	l finance
Allocation of voucher	0.111 (0.072)	N/A	0.111* (0.046)	N/A	0.030 (0.053)	N/A
Take up of the voucher	N/A	0.243 (0.157)	N/A	0.217* (0.090)	N/A	0.059 (0.106)
Obs.	858	858	1,758	1,758	1,316	1,316
	Capability 2	7: Recruiting	Capability	8: Using IT	Share of th	eme-specific
	new en	nployees	to g	grow	practice	s adopted
Allocation of voucher	-0.002 (0.049)	N/A	0.042 (0.044)	N/A	$0.274^{**}$ (0.092)	N/A
Take up of the voucher	N/A	-0.004 (0.095)	N/A	0.081 (0.086)	N/A	$0.536^{**}$ (0.178)
Obs.	1,614	1,614	1,926	1,926	1,965	1,965
	Practice 1	: Having a	Practice 2	: Having a	Practice 3	: Having a
	busine	ss plan	market	ing plan	marketii	ng budget
Allocation of voucher	0.017 (0.028)	N/A	0.024 (0.029)	N/A	0.062* (0.028)	N/A
Take up of the voucher	N/A	0.033 (0.055)	N/A	0.046 (0.056)	N/A	0.120* (0.056)
Obs.	1,952	1,952	1,955	1,955	1,958	1,958
	Practice 4	Practice 4: Having a		: Having a	Practice 6	: Having a
	corporat	e website	workforc	e strategy	recruitme	ent budget
Allocation of voucher	0.030 (0.018)	N/A	0.046 (0.026)	N/A	0.001 (0.020)	N/A
Take up of the voucher	N/A	0.058 (0.036)	N/A	0.089 (0.051)	N/A	0.001 (0.038)
Obs.	1,948	1,948	1,951	1,951	1,959	1,959
	Practice 7	: Having a	Practice 8: H	Having a cash	Practice 9	: Having a
	training	g budget	flow fe	brecast	regular fina	encial review
Allocation of voucher	0.031 (0.027)	N/A	-0.007 (0.026)	N/A	0.037 (0.021)	N/A
Take up of the voucher	N/A	0.060 (0.054)	N/A	-0.013 (0.051)	N/A	0.073 (0.040)
Obs.	1,959	1,959	1,954	1,954	1,955	1,955

Table 6. OLS and IV estimates on capabilities and practices for firms in the 24-month survey

Note: OLS and IV estimates from our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

# 5.3 Extension: heterogeneity by type of diagnostic, theme of advice and firm characteristic

One of the objectives set out in the trial protocol of the programme was to compare the impact of each type of diagnostic (personal or online). To assess the effect of the GVP by diagnostic type, we estimate the ITT and ATT coefficients restricting the sample to the sub-set of firms that received each diagnostic.<sup>29</sup> The implementation of GVP complicates the comparison of the effects by type of diagnostic. First, as discussed above, firms applying in the last month were not randomly allocated to a type of diagnostic. Second, dropout rates differ by diagnostic. Of the 7,743 firms that received diagnostic support but did not complete the application process, 74% were from the personal track and only 26% received an online diagnostic. However, from the total of 20,283 firms that completed the diagnostic and application process, those who received a personal diagnostic and were allocated a voucher, ended up having a higher voucher take up (39% vs. 27% from the online diagnostic track). One possible explanation for these differences is that arranging a face-to-face meeting or a phone call with a delivery partner for the diagnostic was more costly, compared to simply completing an online questionnaire. Given these differences, our estimates for the effect of GVP by type of diagnostic are likely to have different external validity across these two groups. This imbalance urges caution when interpreting these results.

Table A.12 contains the results on firm performance from IDBR. The overall impact of the GVP on turnover after 12 months (8.2%) is no longer significant in any of the groups, although the point estimates are similar in magnitude (8.0% and 8.3% for personal and online diagnostic respectively). This is explained by larger standard errors, which are driven by a smaller sample size in each case. The programme had no effect on employment for either group.<sup>30</sup> For employment, coefficients are negative for personal diagnostic and positive for online diagnostic in all periods. None of these differences are large in comparison to the standard errors.

GVP offered five themes of business advice: sales & marketing, raising finance, leadership & management, workforce development and digital technologies. The trial protocol complicates the interpretation of estimates by theme of advice as the theme is

<sup>&</sup>lt;sup>29</sup> The results are similar when considering the whole sample and including an interaction term between diagnostic type and voucher allocation (for ITT estimates) or voucher take up (for IV estimates).

<sup>&</sup>lt;sup>30</sup> We obtain the same set of results when excluding firms that enrolled in the last month of the application period (the random allocation between type of diagnostic was not followed in this month).

self-selected. For those receiving the personal diagnostic this self-selection occurs because the firm selects the theme when discussing their needs and preferences with the advisor. For firms using the online diagnostic, it occurs because the outcome of the tool is based on their responses to questions in each area. Further, firms could select a theme of advice different to the outcome of the online questionnaire.

With this caveat in mind and given the effects on turnover reported so far, the most interesting estimates by theme are for firms receiving advice on sales & marketing (Table A.13). As before, GVP had no significant impact on employment for this theme of advice but effects on turnover are larger and persistent – the IV estimates indicate turnover increased by 18.8% and 19.8% after 12 and 24 months, respectively.<sup>31</sup> The increase in turnover corresponds to £133,065 and £140,323, respectively, given the average turnover of compliers in 2015. GVP had no impact on turnover for firms which selected any other theme of advice.<sup>32</sup> This suggests the overall GVP impact on turnover after 12 months is driven by firms which selected sales & marketing. While the effect on turnover among those receiving advice on sales & marketing is similar after 24 months, point estimates are smaller for all the other themes of advice, which ultimately results in an overall insignificant effect on turnover after 24 months. There are no significant effects on firm performance for firms that selected raising finance, leadership & management, workforce development or digital technologies.

We also expand the analysis to explore heterogeneity by firm size and age, even though this was not covered in the trial protocol (see section A.3). We do not find clear patterns on the impact of GVP along either of these dimensions. In summary, we find no evidence that the impact of GVP on firm performance differs according to the type of diagnostic received or the characteristic of the firm. Receiving advice on sales & marketing led to a larger and longer lasting effect on turnover, while there was no effect on turnover for firms on other themes.

<sup>&</sup>lt;sup>31</sup> The results are similar when not restricting the sample to the sub-set of firms that selected each theme of advice, but considering the whole sample and including interaction terms for each theme of advice and voucher allocation (for ITT estimates) or voucher take up (for IV estimates).

<sup>&</sup>lt;sup>32</sup> This finding is in line with Anderson, Chandy and Zia (2018) who analyse a randomised controlled trial in South Africa and conclude that marketing training was more successful at increasing turnover than training on finance.

# 6. Comparing RCT to PSM estimates

In addition to providing evidence on the impact of subsidising business advice, a secondary objective for GVP was to provide evidence on the feasibility of experimental designs as a means to improve evaluations of policy effectiveness (BIS, 2016). This section provides such evidence by comparing the RCT-ATT estimates to those obtained from propensity score matching (PSM) – a methodology widely used in policy evaluation as an alternative to trials. Our secondary data has good coverage of both applicants and non-applicants. The randomisation in GVP helps to control for the fact that firms selected into treatment based on unobservable characteristics that may also affect firm performance. PSM can control for selection on observable but not unobservable characteristics and so comparing RCT to PSM illustrates the impact of selection on unobservables.

To implement PSM, we pool firms from all themes of advice, and estimate the ATT by comparing firms that used the voucher (i.e., compliers) to a control group of non-applicant firms matched from the IDBR.<sup>33</sup> The set of non-applicants used for matching is restricted to those in IDBR which meet the eligibility criteria of the programme and are from the same industries as GVP applicants.<sup>34</sup> The propensity score comes from estimating a probit model for the probability a firm receives and uses a voucher based on a rich set of covariates.<sup>35</sup> Each treated firm is matched to three nearest neighbours based on the propensity score restricted to the common support region and the resulting control group used to estimate the ATT. For the assignment equation, we use the probit specification which minimises the number of differences between treated and control groups at baseline after controlling for the propensity score.<sup>36</sup> The PSM estimates use between 1,182 and 1,444 GVP compliers – as with the RCT the number of observations varies by year depending on the number of firms with updated IDBR data for a given outcome and year.<sup>37</sup>

<sup>&</sup>lt;sup>33</sup> PSM estimates exclude non-compliers and non-recipients as they received diagnostic support.

 $<sup>^{34}</sup>$  GVP was open to participants from all industries, but firms did not apply from some sectors. We limit the PSM donor pool to the industries of GVP applicants. This corresponds to 550 4-digit SIC 2007 industries from a total of 724 in IDBR.

<sup>&</sup>lt;sup>35</sup> These include age, age squared, turnover in 2015 and 2014, employment in 2014 and 2015, as well as dummy variables for region, industry and an interaction for London and the biggest sectors; professional, scientific & technological activities, and wholesale & retail.

 $<sup>^{36}</sup>$  Table A.16 and Table A.17 contain five different specifications for the assignment equation of PSM, for turnover and employment, respectively. In each case, the last row (5) corresponds to the model used to obtain the ATT estimate. As shown in these tables, the model considered is the specification which results in the minimum number of differences between treated and control groups at baseline after controlling for the propensity score.

<sup>&</sup>lt;sup>37</sup> As described above, some participants claimed the voucher after the 1st quarter of 2016, which corresponds to the date of IDBR 2016. These are considered as non-participants and excluded in PSM estimates after 12 months, but considered as participants and included in PSM estimates for subsequent periods.

A total sample of more than 394,000 non-applicant firms are used for the matching process. The estimation was conducted with bootstrap (non-parametric with replacement, using 1,000 repetitions) to derive standard errors for the propensity score and the average treatment effect on the treated. Section A.4 provides details.

For each outcome and period, Figure A.1 plots the distribution of the propensity score for GVP compliers and the matched control group. The quality of the match is good and after controlling for the propensity score there are only a few statistically significant differences between treatment and control groups on firm characteristics at baseline (one and two for turnover after 12 and 24 months respectively, and one and two for employment after 24 and 36 months respectively; see Table A.14 and Table A.15).

Table 7 contains ATT estimates from PSM. For all outcomes and periods, the PSM estimates are positive and significant, and increase over time. As can be seen from Table 8, there are substantial differences between the RCT-IV estimates of ATT (column 1) and the PSM estimates (column 2). The RCT results suggest no effect on employment, in contrast to the positive and increasing effects found using PSM. The turnover estimates are reasonably close at 12-months but deviate markedly at 24-months – increasing in the PSM, while converging to zero in the RCT.

	ATT	No. of GVP compliers	Obs.
Turnover after 12 months	0.063** (0.020)	1,363	452,742
Turnover after 24 months	0.125** (0.032)	1,301	416,292
Employment after 12 months	0.031** (0.008)	1,376	480,069
Employment after 24 months	0.077** (0.012)	1,276	432,989
Employment after 36 months	0.100** (0.018)	1,182	394,644

Table 7. PSM estimates for turnover and employment

Note: PSM estimates for percentage changes since baseline. Bootstrapped standard errors in parentheses; \*\* p<0.01, \* p<0.05.

The PSM estimates in column (2) capture the joint effect of i) selection bias; ii) receiving diagnostic support; iii) subsidised business advice. We can exploit the fact that GVP vouchers were randomly allocated to provide an estimate of the joint effect of (i) and (ii).<sup>38</sup> These effects are reported in column 3 of Table 8 and are estimated by using PSM to

<sup>&</sup>lt;sup>38</sup> We cannot isolate the effect of selection since all GVP applicants received diagnostic support.

compare GVP applicants who received the diagnostic but not the voucher against a control group of firms constructed using eligible non-applicants.<sup>39</sup> These estimates can be used to isolate the effect of subsidised business advice by taking the difference between column 2 and column 3, as reported in column 4. Comparing to column (1) we see that these PSM estimates of the ATT now match those reported in column 1 in terms of magnitude and significance.

In summary, PSM provides highly misleading results for turnover and employment. Excluding the effect of diagnostic and selection using the randomisation in GVP – which would, of course, not be available in a non-trial setting – PSM estimates are similar to IV. Despite the availability of high-quality secondary data on a range of observables, unobservable characteristics that influence the selection of firms into treatment introduce a large upward bias to PSM estimates, which increases over time.<sup>40</sup>

able 0. Comparison of 1 v and	A'	TT	PSM effect of GVP	PSM effect of
	IV	PSM	selection and diagnostic	GVP subsidised advice (2-3)
	(1)	(2)	(3)	(4)
Turnover after 12 months	0.082*	0.063**	-0.004	0.067**
	(0.039)	(0.020)	(0.020)	(0.015)
Turnover after 24 months	0.024	0.125**	0.088**	0.037
	(0.061)	(0.032)	(0.032)	(0.023)
Employment after 12 months	-0.026	0.031**	0.032**	-0.001
	(0.020)	(0.008)	(0.009)	(0.007)
Employment after 24 months	-0.022	0.077**	0.066**	0.011
	(0.032)	(0.012)	(0.014)	(0.011)
Employment after 36 months	-0.000	0.100**	$0.095^{**}$	0.005
	(0.043)	(0.018)	(0.018)	(0.015)

Table 8. Comparison of IV and PSM estimates

Note: IV and PSM estimates for percentage changes since baseline. Robust standard errors in parentheses for column 1 and bootstrapped standard errors in parentheses for columns 2 and 3. Standard errors of column 4 obtained from a Wald test for the difference between the coefficient of dummy variable for compliers and the coefficient of dummy variable for non-recipients, both derived from an OLS regression for each outcome of interest, controlling for the propensity score and with robust standard errors; \*\* p<0.01, \* p<0.05.

# 7. Conclusions

GVP was set up as a randomised controlled trial and aimed to encourage firms to take-up business advice and to help them grow. The programme increased turnover, but not

<sup>&</sup>lt;sup>39</sup> These PSM estimates exclude compliers and non-compliers.

<sup>&</sup>lt;sup>40</sup> Michalopoulos, Bloom and Hill (2004) also conclude that the bias is smaller in the short-term, when comparing experimental vs. PSM estimates.

employment, among treated firms. The impact on turnover was temporary – turnover increased 8.2% but only after the first 12 months with no significant effect at 24 months. Changes in self-reported outcomes, capabilities and practices are all broadly consistent with these headline findings. Survey estimates suggest that development of new products, rather than improved access to the UK or other markets could explain growth in turnover after 12 months.

Our results on firm performance, capabilities, and practices are consistent with other evaluations.<sup>41</sup> That is: i) positive effects for some outcomes, not others; ii) results that are somewhat better for turnover than employment; iii) effects that tend to be temporary (see, for example, Nagel, Huber, Van Praag & Goslinga, 2019); and iv) some improvements in capabilities and practices that are not reflected on firm performance.<sup>42</sup>

While it is tempting to speculate why the effect is temporary – perhaps GVP failed to address barriers to financing physical capital (Kosters & Obschonka, 2011; Rogers & Helmers, 2008; Schoonjans et. al. 2013) or improve managerial capital (Bertrand and Schoar, 2003; Bennedsen et al. 2007; Bloom and van Reenen, 2010); or, more generally, because the advice on longer term changes was of poor quality or insufficient, or not implemented by firms, GVP provides no additional evidence to distinguish between these competing hypotheses.<sup>43</sup>

Nevertheless, the impact on turnover is large, and while these gains in turnover are also large relative to costs, one concern with this effect is the extent to which it represents displacement.<sup>44</sup> Somewhere between 69% and 76% of those that received and used the voucher are firms in non-tradable sectors and medium tradable services – i.e., they are likely to serve local markets. Consistent with this, only 25% of firms are exporting at baseline and GVP has no effect on starting exporting (both statistics come from self-

<sup>&</sup>lt;sup>41</sup> For a systematic review of evaluations from OECD countries, see http://www.whatworksgrowth.org/policy-reviews/business-advice/.

<sup>&</sup>lt;sup>42</sup> Lambrecht & Pirnay (2005) find no effect on performance from subsidised private external consultancy in Belgium, but improvements on management practices and capabilities. Bardasi et al. (2021) and Karlan, Knight & Udry (2015) find that business advice leads to the adoption of new and better business practices (in Tanzania and Ghana respectively) but find no effect on revenues or profits. Iacovone, Maloney & Mckenzie (2019) compare two types of business support in Colombia; both improve management practices, yet only one of them leads to a positive effect on employment.

 $<sup>^{43}</sup>$  The quality and intensity of the business advice, and the extent to which firms followed it, was not monitored during the trial.

<sup>&</sup>lt;sup>44</sup> Although these effects are large, other quasi-experimental evaluations find even larger effects. For example, Manaresi et al. (2022) evaluating an export voucher subsidy for SMEs in Italy find larger impacts on multiple outcomes and periods.

reported outcomes in survey data).<sup>45</sup> This suggests that a significant proportion of the increased sales at GVP supported firms could be at the expense of other non-supported firms, although the data available do not allow us to quantify the magnitude of this substitution effect.

One objective of GVP, being implemented as a randomised controlled trial, was to improve the robustness of evaluations and provide evidence in favour of experimental designs for assessing policy effectiveness – these results are consistent with that objective. Despite the availability of comprehensive secondary data and achieving good matching, PSM provides highly misleading results for turnover and employment. Comparing results from PSM to those from the RCT, we show that the bias introduced from unobservables is not only large, but it increases over time.

The main rationale for governments subsidising business advice is that many firms underinvest in business support due to market failures, and that firms would perform better by improving their management and production processes. The programme had a positive and large effect on turnover, albeit potentially at the expense of other nonsupported local firms. The subsidised advice appears to have improved firms' capabilities and practices in a way that was consistent with the increase in turnover. However, the effect on turnover did not persist beyond 12 months. Further evidence is needed to better identify how the kind of business advice provided through GVP, or similar programmes, can generate persistent effects on firm performance. Our finding that results from PSM can be highly misleading suggests that additional randomised controlled trials will play a key part in developing this evidence.

<sup>&</sup>lt;sup>45</sup> Non-tradable corresponds to the public sector, construction, and the least tradable services, following Faggio & Overman (2014) and Jensen & Kletzer (2006). See appendix section A.5.

# References

Abadie, A., & Imbens, G. W. (2006). Large sample properties of matching estimators for average treatment effects. *Econometrica*, 74(1), 235-267.

Agodini, R. and Dynarski, M. (2004). "Are Experiments the Only Option? A Look at Dropout Prevention Programs", *The Review of Economics and Statistics*, vol.86(1), pp.180-194. Akerlof, G. (1970). The market for lemons. Quarterly journal of Economics, 84(3), 488-500.

Anderson, S. J., Chandy, R., & Zia, B. (2018). Pathways to profits: The impact of marketing vs. finance skills on business performance. *Management Science*, *64*(12), 5559-5583.

Angrist, J. D., & Pischke, J. S. (2009). Instrumental variables in action: sometimes you get what you need. Mostly harmless econometrics: an empiricist's companion, 113-220.

Angrist, J. D., Imbens, G. W., & Rubin, D. B. (1996). Identification of causal effects using instrumental variables. *Journal of the American statistical Association*, 91(434), 444-455.

Bardasi, E., Gassier, M., Goldstein, M., & Holla, A. (2021). The Profits of Wisdom: The Impact of a Business Support Program in Tanzania. *The World Bank Economic Review*, 35(2), 328-347.

Bartle, D., and Morris, M. (2010) Evaluating the impacts of government business assistance programmes: approaches to testing additionality. *Research Evaluation*, 19, 275–280.

Bennedsen, M., Nielsen, K. M., Pérez-González, F., & Wolfenzon, D. (2007). Inside the family firm: The role of families in succession decisions and performance. *The Quarterly Journal of Economics*, *122*(2), 647-691.

Bennett, R. (2008). SME policy support in Britain since the 1990s: what have we learnt? *Environment and Planning C: Government and Policy*, 26(2), 375-397.

Benus, J., Shen, T., Zhang, S., Chan, M. and Hansen, B. (2010) Growing America Through Entrepreneurship: Final Evaluation of Project GATE. Washington DC, US Department of Labor Employment and Training Administration.

Benus, J.M., Wood, M.L., Grover, N., and Abt Associates (1994) A Comparative Analysis of the Washington and Massachusetts UI Self-Employment Demonstrations. Cambridge, Mass, Abt Associates.

BERR (2006); "Economic Impact Study of Business Link Local Service", K. Mole, M. Hart, S. Roper, D. Saal, D. Storey; UK Department for Business, Enterprise and Regulatory Reform

Bertrand, M., & Schoar, A. (2003). Managing with style: The effect of managers on firm policies. *The Quarterly journal of economics*, *118*(4), 1169-1208.

Bloom, N., & Van Reenen, J. (2010). Why do management practices differ across firms and countries? *Journal of economic perspectives*, 24(1), 203-24.

Bloom, N., Sadun, R., & Van Reenen, J. (2012). Americans do IT better: US multinationals and the productivity miracle. *American Economic Review*, 102(1), 167-201

Breinlich, H., Mion, G., Nolen, P., and Novy, D. (2012) Intellectual Property, Overseas Sales, and the Impact of UKTI Assistance in Entering New Overseas Markets (UKTI). London, UKTI.

Bruhn, M., & Zia, B. (2013). Stimulating managerial capital in emerging markets: the impact of business training for young entrepreneurs. *Journal of Development Effectiveness, 5*(2), 232-266.

Bruhn, M., Karlan, D.S., and Schoar, A. (2018) The Impact of Consulting Services on Small and Medium Enterprises: Evidence from a Randomized Trial in Mexico. *Journal of Political Economy*, *126*(2), 635-687.

Business Innovation & Skills (BIS). (2013). "SMEs: The key enablers of business success and the economic rationale for government intervention". BIS Analysis Paper Number 2.

(2014). "Growth Vouchers Programme: Formative Evaluation Report". BIS Research Paper Number 219.

Qualitative Assessment Report". BIS Research Paper Number 220.

(2016). "Exploring the feasibility of a productivity based approach for evaluating business support interventions". BIS Research Paper Number 295.

Caliendo, M., & Kopeinig, S. (2008). Some practical guidance for the implementation of propensity score matching. *Journal of economic surveys*, 22(1), 31-72.

Camuffo, A., Cordova, A., Gambardella, A., & Spina, C. (2020). A scientific approach to entrepreneurial decision making: Evidence from a randomized control trial. *Management Science*, 66(2), 564-586.

Cumming, D.J., and Fischer, E. (2012) Publicly Funded Business Advisory Services and Entrepreneurial Outcomes. Research Policy 41, 467–481.

Criscuolo, C., R. Martin, et al. (2012); "The Causal Effects of an Industrial Policy", NBER Working Paper Series Working Paper 17842.

De Loecker, J., Obermeier, T. and Van Reenen, J. (2022), 'Firms and inequality', *IFS Deaton Review of Inequalities*, https://ifs.org.uk/inequality/firms-and-inequality.

Dehejia, R. H., and Wahba, S. (1999). "Causal Effects in Nonexperimental Studies: Reevaluating the Evaluation of Training Programs." *Journal of the American Statistical Association*, 94: 1053-62.

Dehejia, R. H., & Wahba, S. (2002). Propensity score-matching methods for nonexperimental causal studies. *Review of Economics and statistics*, 84(1), 151-161.

Diaz, J.J. and Handa, S. (2006). "An Assessment of Propensity Score Matching as a Nonexperimental Impact Estimator: Evidence from Mexico's PROGRESA program", *The Journal* of Human Resources, vol. 41(2), pp. 319-345.

Drews, C., and Hart, M. (2015) "Feasibility Study – Exploring the Long-Term Impact of Business Improvement Services" Enterprise Research Centre Research Paper No. 29, March 2015.

Faggio, Giulia & Overman, Henry, 2014. "The effect of public sector employment on local labour markets," Journal of Urban Economics, Elsevier, vol. 79(C), pages 91-107.

Fairlie, R. W., Karlan, D., & Zinman, J. (2015). Behind the GATE experiment: Evidence on effects of and rationales for subsidized entrepreneurship training. *American Economic Journal: Economic Policy*, 7(2), 125-61.

Fraker, T. and Maynard, R. (1987). "The adequacy of comparison group designs for evaluations of employment-related programs", *The Journal of Human Resources*, vol.22(2), pp.194-227.

Georgiadis, A., & Pitelis, C. N. (2016). The impact of employees' and managers' training on the performance of small-and medium-sized enterprises: Evidence from a randomized natural experiment in the UK service sector. *British Journal of Industrial Relations*, 54(2), 409-421.

Girma, S., Görg, H., and Pisu, M. (2005). Quantitative analysis and linked micro-data study of UKTI services - Final report. London, UKTI.

Godel, M., and Mantovani, I. (2012) Evaluation of the impact and cost effectiveness of UKTI's regional network support. London, UKTI.

Grossman, S. J., & Stiglitz, J. E. (1980). On the impossibility of informationally efficient markets. *The American economic review*, 70(3), 393-408.

Hart, M., and Roper, S. (2003) Modelling the Effects of Public Support to Small Firms in the UK - Paradise Gained? *European Regional Science Association*. Jyväskylä, Finland, August 2003.

Heckman, J., Ichimura, H., Smith, J. and Todd, P. (1996). "Sources of selection bias in evaluating social programs: an interpretation of conventional measures and evidence on the effectiveness of matching as a program evaluation method." Proceedings of the National Academy of Sciences 93(23), 13416-13420.

Heckman, J. J., Ichimura, H., & Todd, P. E. (1997). Matching as an econometric evaluation estimator: Evidence from evaluating a job training programme. *Review of economic studies*, *64*(4), 605-654.

Heckman, J., Ichimura, H., Smith, J., & Todd, P. (1998). "Characterising selection bias using experimental data." *Econometrica* 66(5): 1017-1098.

Heckman, J. J., Urzua, S., & Vytlacil, E. (2006). Understanding instrumental variables in models with essential heterogeneity. *The Review of Economics and Statistics*, 88(3), 389-432.

Heckman, J. J., & Vytlacil, E. J. (1999). Local instrumental variables and latent variable models for identifying and bounding treatment effects. Proceedings of the national Academy of Sciences, 96(8), 4730-4734.

Heckman, J. J., & Vytlacil, E. (2005). Structural equations, treatment effects, and econometric policy evaluation 1. *Econometrica*, *73*(3), 669-738.

Iacovone, L., Maloney, W. F., & Mckenzie, D. J. (2019). Improving management with individual and group-based consulting: Results from a randomized experiment in Colombia. *World Bank Policy Research Working Paper*, (8854).

Imbens, G. W. (2004). Nonparametric estimation of average treatment effects under exogeneity: A review. *Review of Economics and statistics*, 86(1), 4-29.

Jensen, J. B., & L.G. Kletzer, L.G. (2006). "Tradable services: understanding the scope and impact of services outsourcing" S.M. Collins, L. Brainard (Eds.), Brookings Trade Forum 2005, Offshoring White-Collar Work, Brookings Institution, Washington DC, pp. 75-134.

Karlan, D., Knight, R., & Udry, C. (2012). Hoping to win, expected to lose: Theory and lessons on micro enterprise development (No. w18325). *National Bureau of Economic Research*.

Karlan, D., Knight, R., & Udry, C. (2015). Consulting and capital experiments with microenterprise tailors in Ghana. *Journal of Economic Behavior & Organization*, 118, 281-302.

Karlan, D., & Valdivia, M. (2011). Teaching entrepreneurship: Impact of business training on microfinance clients and institutions. *Review of Economics and statistics*, 93(2), 510-527.

Kosters, S., and Obschonka, M. (2011) Public Business Advice in the Founding Process: An Empirical Evaluation of Subjective and Economic Effects. *Environment and Planning C*, 29, 577–604.

LaLonde, R. J. (1986). "Evaluating the Econometric Evaluations of Training Programs Using Experimental Data." *American Economic Review*, 76: 602-20.

Lambrecht, J., & Pirnay, F. (2005). An evaluation of public support measures for private external consultancies to SMEs in the Walloon Region of Belgium. *Entrepreneurship & Regional Development*, 17(2), 89-108.

Lechner, M. (2001). Identification and estimation of causal effects of multiple treatments under the conditional independence assumption. In Econometric evaluation of labour market policies (pp. 43-58). Physica, Heidelberg.

Leuven, E., & Sianesi, B. (2018). PSMATCH2: Stata module to perform full Mahalanobis and propensity score matching, common support graphing, and covariate imbalance testing.

Manaresi F., Palma, A., Salvatici, L. and Scrutinio V. (2022). Managerial Input and Firm Performance. Evidence from a Policy Experiment. Istituto Nazionale Previdenza Sociale. WorkINPS Papers. June 2022 – No. 51. Mano, Yukichi, AlHassan Iddrisu, Yutaka Yoshino and Tetsushi Sonobe (2012) "How Can Micro and Small Enterprises in Sub-Saharan Africa Become More Productive? The Impacts of Experimental Basic Managerial Training", World Development, 40(3): 458-68.

sMcKenzie, D., & Woodruff, C. (2014). What are we learning from business training and entrepreneurship evaluations around the developing world? *The World Bank Research Observer,* 29(1), 48-82.

Mansson, J., and Widerstedt, B. (2012). The Swedish Business Development Program: Evaluation and some methodological and practical notes. European Regional Science Association, European Society of Regional Analysis. Bratislava, Slovakia, 2012.

Michalopoulos, C., Bloom, H.S. and Hill, C. J. (2004). "Can Propensity-Score Methods Match the Findings from a Random Assignment Evaluation of Mandatory Welfare-to-Work Programs?", *The Review of Economics and Statistics*, vol.86(1), pp.156-179.

Mole, K. F., & Keogh, W. (2009). The implications of public sector small business advisers becoming strategic sounding boards: England and Scotland compared. *Entrepreneurship and regional development*, 21(1), 77-97.

Nagel, H., Huber, L. R., Van Praag, M., & Goslinga, S. (2019). "The effect of a tax training program on tax compliance and business outcomes of starting entrepreneurs: Evidence from a field experiment." *Journal of business venturing*, 34(2), 261-283.

Nannicini, T. (2007). Simulation-based sensitivity analysis for matching estimators. *Stata Journal*, 7(3), 334.

North, D., Baldock, R., Mole, K., Wiseman, J., & Binnie, C. (2011). Research to understand the barriers to take up and use of business support. London: For the Department for Business Innovation and Skills.

Oberschachtsiek, D., Scioch, P. (2011) The outcome of coaching and training for selfemployment: a statistical evaluation of non-financial support schemes for unemployed business founders in Germany. Nuremberg, Institute for Employment Research of the Federal Employment Agency (IAB).

OECD (2002) OECD Small and Medium Enterprise Outlook, OECD, Paris.

Office for National Statistics. (2019). Business Structure Database, 1997-2018: Secure Access. 10th Edition. UK Data Service.

Polanyi, M. T. (1966). The tacit dimension. London: Routledge & Kegan Paul.

Powell, T. 1995. Total quality management as competitive advantage: A review and empirical study. *Strategic Management Journal, 16*: 15–37.

Robson, P.J.A., and Bennett, R.J. (2000) SME Growth: The Relationship with Business Advice and External Collaboration. *Small Business Economics*, 15, 193–208.

Rogers, EM. 1995. Diffusion of Innovations, New York: Free Press.

Rogers, M., and Helmers, C. (2008) Intellectual property and UKTI Passport firms - Stage 2 Report. London, UKTI.

Romano, J. P., and M. Wolf. 2005a. Exact and Approximate Stepdown Methods for Multiple Hypothesis Testing. *Journal of the American Statistical Association* 100(469): 94–108.

*Econometrica* 73(4): 1237–1282.

Sanders, C. (2002) The Impact of Microenterprise Assistance Programs: A Comparative Study of Program Participants, Nonparticipants, and Other Low-Wage Workers. *Social Science Review*, 76, 321-340.

Schoonjans, B., Cauwenberge, P.V., and Bauwhede, H.V. (2013) Knowledge networking and growth in service firms. *The Service Industries Journal*, 33 (11), 1051-1067.

Smith, J. A., & Todd, P. E. (2001). Reconciling conflicting evidence on the performance of propensity-score matching methods. *American Economic Review*, 91(2), 112-118.

Smith, J. A., and Todd, P. E. (2005). "Does Matching Overcome LaLonde's Critique of Nonexperimental Estimators." *Journal of Econometrics*, 125: 305-53.

Solon, G., Haider, S. J., & Wooldridge, J. M. (2015). What are we weighting for? Journal of Human resources, 50(2), 301-316.

Stiglitz, J. E. (1979). Equilibrium in product markets with imperfect information. *American Economic Review*, 69(2), 339-345.

Stock J, Yogo M, Wright J. A Survey of Weak Instruments and Weak Identification in Generalized Method of Moments. *Journal of Business and Economic Statistics*. 2002;20:518-529.

Valdivia, M. (2015). Business training plus for female entrepreneurship? Short and mediumterm experimental evidence from Peru. *Journal of Development Economics*, *113*, 33-51.

Van Cauwenberge, P., Vander Bauwhede, H., and Schoonjans, B. (2013). An evaluation of public spending: the effectiveness of a government-supported networking program in Flanders. *Environment and Planning C*, 31, 24–38.

Weidmann, B. and Miratrix, L. (2021). Lurking Inferential Monsters? Quantifying Selection Bias in Evaluations of School Programs. *Journal of Policy Analysis and Management*, 40: 964-986.

Wilde, E.T. and Hollister, R. (2007). "How close is close enough? Evaluating propensity score matching using data from a class size reduction experiment", *Journal of Policy Analysis and Management*, vol.26, pp.455-477.

Wren, C., & Storey, D. (2002). Evaluating the Effect of Soft Business Support upon Small Firm Performance. Oxford Economic Papers, 54(2), 334-365. Retrieved April 22, 2020.

# Appendix

## A.1 Data

#### Variable construction and processing

- Employment: IDBR includes number of employees and employment - where the latter is constructed by adding the number of proprietors to the former, which for many firms correspond to imputed data. We use number of employees for the analysis, to limit the measurement error that can arise from this imputed data.

- Winsorising: For employment and turnover, we obtained the annual percentage change and winsorise the top 1% of the distribution across the whole sample, including nonapplicants. For employment, we also scaled the values by 1, to avoid having firms with zero employees.

- Survival: IDBR includes the death year of the firm, which we use to construct a dummy variable equal to one whenever the firms is alive and zero otherwise.

#### Sample restrictions

- We excluded 25 firms with repeated unique identifiers.

- Through IDBR we identified a small sub-set of firms that were not eligible to participate in the programme but that ultimately applied and were included (less than 0.8% of completed applications). For these, we introduced two set of rules applied to our analysis from both IDBR and survey data; i) we excluded firms that started trading in 2015 or later, those that stopped operating in 2012 or before and firms with more than 60 employees in 2014; and ii) we included firms that had between 50 and 60 employees in 2014 and those from regions in the UK beyond England, which were not originally covered by GVP.<sup>46</sup>

#### Timing of survey and secondary data

For the survey data, the timing of interviews was relative to the date of diagnostic. In contrast, IDBR provides us with a 'point-in-time' snapshot regardless of the date of diagnostic. Using the same survey-cohorts for the analysis with IDBR data would

<sup>&</sup>lt;sup>46</sup> According to the IDBR, 71 firms started trading in 2015 or later (63 were allocated a voucher); 29 stopped operating in 2012 or before (20 were allocated a voucher), 33 had more than 60 employees in 2014 (24 were allocated a voucher), 15 had between 50 and 60 employees in 2014 (12 were allocated a voucher), and 13 were registered in a region not originally covered by GVP (12 were allocated a voucher).

introduce up to 12 months variation in the length of time that had passed post-diagnostic and post-treatment. Instead, for the analysis using the IDBR we create our own cohorts based on the date of diagnostic relative to the date of the IDBR. We use the date of diagnostic rather than the voucher claim date as the latter is only available for a sub-set of firms.

To do this, we annualize the IDBR data using cut-offs 6 months before and after the midpoint of IDBR in each year. As the midpoint of IDBR is the 15<sup>th</sup> of February, this gives us a year-end of 16<sup>th</sup> August. To construct the cohorts, we assign firms to this annualised data using the diagnostic date plus 103 days (median delay diagnostic to voucher claim) against these periods to define the cohort of firms. The first cohort is composed of firms whose diagnostic plus 103 days falls between the 10<sup>th</sup> of May 2014 (the first diagnostic we observe) and 16<sup>th</sup> August 2014, the second cohort is composed of firms for whom this date is between the 17<sup>th</sup> August 2014 and the 5<sup>th</sup> of August 2015 (the last diagnostic we observe).<sup>47</sup> For the analysis of IDBR outcomes we consider the month of application and IDBR-cohort of firms as our set of basic control variables.

We use IDBR 2014 for the baseline for the first cohort, and IDBR 2015 for the second cohort. This raises the possibility that some firms in cohort 2 will have received advice just before baseline. At most, firms that receive their diagnostics in mid-August and spend their voucher immediately will receive advice 6 months before baseline. Yet, for the average firm, having the diagnostic in mid-August implies receiving advice only 2.5 months before baseline. We do not worry about this since even 6 months is a relatively short period for introducing changes within the business and improving firm performance. Besides, the reporting of IDBR has some lags and considering the following IDBR year for these firms allows us to reduce data gaps and increases our sample size. The final three rows of Table 1 show the results of constructing IDBR cohorts in this way. For the first cohort, IDBR data for 2015 provides outcomes on average 9.7 months after the voucher was claimed. Among firms that received and used the voucher, a median of 103 days passed between the date at which they received their diagnostic and when they claimed the voucher. This means that on average, the 12-month IDBR reports outcomes somewhere between 9.7 and 13.1 months after treatment. Similarly, on average the 24-

<sup>&</sup>lt;sup>47</sup> The resultant allocation of firms to cohorts corresponds to 1,345 assigned to the 2015 cohort (74.5% allocated a voucher and 25.5% from the control group), while 12,099 firms to the 2016 cohort (75.0% allocated a voucher and 25.0% from the control group). In addition, the cohort was imputed for 13 firms with missing information for the diagnostic date; these were allocated to the 2016 cohort.

month IDBR reports outcomes between 21.7 and 25.1 months after treatment, and the 36-month IDBR reports outcomes between 33.7 and 37.1 months after.

To illustrate this, the first two rows of Table 1 report statistics for the timing of the survey relative to the date at which the firms claimed their voucher. Firms claimed the voucher after receiving the business advice, so this date marks the end of their participation in the programme.<sup>48</sup> We do not know how quickly firms claimed the voucher following the receipt of advice. We only know that the treatment occurred sometime between the date of the diagnostic and the date at which the voucher was claimed. The rule, not always enforced, was that vouchers had to be claimed within 90 days from the date of diagnostic. The first row of Table 1 shows that the 12-month survey reports data that, on average, comes only 7.9 months after the voucher was claimed. Not all firms used the voucher, but for those that did around 103 days passed between the date at which they received their diagnostic and when they claimed the voucher.<sup>49</sup> This means that on average, the 12-month survey data reports outcomes somewhere between 7.9 and 11.3 months after treatment. Similarly, on average the 24-month survey reports outcomes between 19.6 and 23 months after treatment. There is considerable variation around these means in both surveys.

<sup>&</sup>lt;sup>48</sup> In order to claim the voucher, businesses had to provide details of the advice received and evidence that an invoice was paid.

<sup>&</sup>lt;sup>49</sup> The median number of days between diagnostic and claim date for firms that received and used the voucher correspond to 103, with an average of 106 days.

# A.2 Additional results, tables, and figures

Firm characteristic in 2015	Obs.	Mean	s.d.	Min.	Max.
Age (no. of years)	13,385	9.4	9.2	1	42
Turnover (thousands of annual £)	10,200	772.5	1,490.0	1	38,531
Turnover % growth (2014 to 2015)	9,148	47.5	920.0	-99.9	69,500
Employment (no. of employees)	8,381	7.3	8.7	1	301
Exporting firm (share of firms)	13,444	0.25	0.4	0	1
Location					
East of England	13,444	0.08	0.28	0	1
East of Midlands	13,444	0.06	0.24	0	1
Greater London	13,444	0.18	0.39	0	1
North (East & West)	13,444	0.16	0.36	0	1
South East	13,444	0.09	0.29	0	1
South West	13,444	0.11	0.31	0	1
West Midlands	13,444	0.14	0.34	0	1
Yorkshire & Humberside	13,444	0.10	0.30	0	1
Other	13,444	0.08	0.27	0	1
Sector (SIC 2007)					
Agriculture, forestry & fishing	13,389	0.01	0.11	0	1
Manufacturing	13,389	0.09	0.28	0	1
Construction	13,389	0.07	0.26	0	1
Wholesale & retail trade	13,389	0.18	0.39	0	1
Transportation & Storage	13,389	0.02	0.13	0	1
Accommodation & food services	13,389	0.03	0.18	0	1
Information & communication	13,389	0.12	0.32	0	1
Real estate activities	13,389	0.02	0.13	0	1
Professional, scientific & technical	13,389	0.23	0.42	0	1
Administrative & support service	13,389	0.10	0.30	0	1
Education	13,389	0.03	0.17	0	1
Other service activities	13,389	0.11	0.31	0	1
Activities of households as employers	13,389	0.01	0.01	0	1

Table A.1. Summary statistics at baseline for GVP firms in IDBR

Note: Age, turnover and employment obtained from IDBR. Exporting status, location and sector obtained from baseline information.

	With covariates	Obs.		With covariates	Obs.
	-0.242	-			-
Age (no. of years)	(0.186)	13,385	Sector		
Turnover $(1,000,f)$	1.464	10 200	Aerospace	-0.002	13/1/1
1 uniover (1,000 2)	(31.561)	10,200	Aerospace	(0.001)	10,444
No. of employees	$-0.568^{**}$	8,381	Agriculture	-0.001	13,444
	0.012			0.002	
Exporting firm (share)	(0.009)	13,444	Automotive	(0.003)	13,444
	-16.989	0 1 4 0	<b>D</b> · ·	0.017**	10.444
Turnover growth (%)	(23.918)	9,148	Business services	(0.006)	13,444
Growth expectations	-0.002	13 360	Clothing/footwoor	-0.001	19 111
Growth expectations	(0.011)	15,505	Clothing/100twear	(0.003)	10,444
Location			Communications	-0.004	13.444
	0.007			(0.003)	- )
East of England	(0.005)	13,444	Construction	(0.003)	13,444
	-0.003			0.000	
East of Midlands	(0.005)	13,444	Creative media	(0.002)	13,444
	-0.001	10.444		0.004	10.444
Greater London	(0.008)	13,444	Education & training	(0.004)	13,444
North (East & Wost)	0.012	13/1/	Floetropics	0.003	13 ///
Northi (Last & West)	(0.007)	10,111	Licetronics	(0.003)	10,111
South East	-0.006	13,444	Environment	0.001	13,444
	(0.006)			(0.002)	
South West	(0.006)	13,444	Food & drink	(0.008)	13,444
	-0.009			-0.000	
West Midlands	(0.007)	13,444	Financial services	(0.001)	13,444
Vorkshing & Humbonsida	-0.009	19 ///	Fine & committy	-0.004	19 111
Torkshire & fruinberside	(0.006)	10,444	File & security	(0.005)	10,444
Other	0.004	13.444	Jewellerv/tableware	0.001	13.444
	(0.005)	- )	· · · · · · · · · · · · · · ·	(0.002)	- )
Capabilities			Healthcare	-0.007	13,444
	0.000			-0.000	
1: Managing people	(0.017)	13,197	Household goods	(0.003)	13,444
	0.002	10.004	T . 0 / .	-0.001	10.444
2: Developing a business plan	(0.019)	13,364	Leisure & tourism	(0.003)	13,444
3. Accessing the UK market	0.011	19 708	Machanical/alactric	0.002	19 111
5. Accessing the OK market	(0.020)	12,700	Wiedhameal/electric	(0.002)	10,444
4: Accessing other markets	0.005	10.684	Oil & gas	0.000	13.444
	(0.027)	,		(0.001)	
5: Developing new products	(0.012)	13,151	Ports & logistics	-0.000	13,444
	0.020)			(0.001)	
6: Accessing external finance	(0.032)	12,087	Computer services	(0.002)	13,444
	0.000	10.010	G + 0.1 ·	0.001	10
7. Recruiting new employees	(0.020)	12,912	Sports & leisure	(0.003)	13,444
8. Using IT to grow	-0.013	13 226	Othors	-0.024**	13 ///
o. Using 11 to grow	(0.024)	10,000	Others	(0.009)	10,444

Table A.2. Balance of treatment and control groups at baseline for firms in IDBR With Old With Old With

Note: Each row corresponds to an OLS regression of voucher allocation on a firm characteristic. The covariates correspond to month of application, cohort and the full set of interactions between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

	With	Oba		With	Oba
	covariates	Obs.		covariates	Obs.
Age (no. of years)	0.172 (0.132)	5,579	Sector		
Self-reported turnover (£)	40,543.6 (25,458.0)	5,579	Aerospace	-0.001 (0.002)	5,579
Self-reported no. of employees	0.010 (0.244)	5,579	Agriculture	-0.003 (0.004)	5,579
Exporting firm (share)	0.006 (0.014)	5,579	Automotive	-0.006 (0.005)	5,579
Having a business plan	-0.025 (0.016)	5,579	Business services	0.013 (0.008)	5,579
Growth expectations	-0.030 (0.018)	5,550	Clothing/footwear	-0.003 (0.005)	5,579
Location			Communications	0.005 (0.004)	5,579
East of England	-0.007 (0.009)	5,579	Construction	0.007 (0.008)	5,579
East of Midlands	-0.001 (0.008)	5,579	Creative media	-0.004 (0.009)	5,579
Greater London	-0.022 (0.013)	5,579	Education & training	-0.007 (0.009)	5,579
North (East & West)	0.009 (0.011)	5,579	Electronics	0.003 (0.005)	5,579
South East	0.007 (0.009)	5,579	Environment	-0.001 (0.004)	5,579
South West	0.010 (0.010)	5,579	Food & drink	0.007 (0.006)	5,579
West Midlands	0.007 (0.011)	5,579	Financial services	0.001 (0.002)	5,579
Yorkshire & Humberside	-0.012 (0.010)	5,579	Fire & security	-0.015* (0.008)	5,579
Other	0.009 (0.008)	5,579	Jewellery/tableware	0.004 (0.003)	5,579
Capabilities			Healthcare	-0.011 (0.007)	5,579
1: Managing people	0.045 (0.027)	5,423	Household goods	-0.001 (0.004)	5,579
2: Developing a business plan	0.039 (0.030)	5,552	Leisure & tourism	0.005 (0.004)	5,579
3: Accessing the UK market	0.033 (0.032)	5,249	Mechanical/electric	0.004 (0.003)	5,579
4: Accessing other markets	0.057 (0.042)	4,410	Oil & gas	-0.001 (0.002)	5,579
5: Developing new products	0.052 (0.032)	5,449	Ports & logistics	-0.001 (0.001)	5,579
6: Accessing external finance	0.034 (0.038)	4,939	Computer services	-0.000 (0.007)	5,579
7: Recruiting new employees	0.046 (0.032)	5,246	Sports & leisure	-0.006 (0.005)	5,579
8: Using IT to grow	0.033 (0.038)	5,535	Others	0.002 (0.013)	5,579

Table A.3. Balance of treatment and control groups at baseline for firms interviewed in the 12-month survey

Note: Each row corresponds to an OLS regression of voucher allocation on a firm characteristic. The covariates correspond to month of application, cohort and the full set of interactions between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. Robust standard errors in parentheses: \*\* p<0.01, \* p<0.05.

	With	Oba		With	Oha
	covariates	Obs.		covariates	Obs.
Age (no. of years)	0.414 (0.239)	1,965	Sector		
Self-reported turnover (£)	72,970.8 (43,148.1)	1,965	Aerospace	0.002 (0.003)	1,965
Self-reported no. of employees	0.581 (0.417)	1,965	Agriculture	-0.004 (0.007)	1,965
Exporting firm (share)	0.012 (0.026)	1,965	Automotive	-0.006 (0.009)	1,965
Having a business plan	-0.061* (0.028)	1,965	Business services	0.042** (0.014)	1,965
Growth expectations	-0.019 (0.031)	1,956	Clothing/footwear	0.006 (0.008)	1,965
Location			Communications	-0.004 (0.008)	1,965
East of England	-0.034 (0.018)	1,965	Construction	0.027* (0.012)	1,965
East of Midlands	0.002 (0.014)	1,965	Creative media	-0.009 (0.017)	1,965
Greater London	0.021 (0.021)	1,965	Education & training	-0.014 (0.016)	1,965
North (East & West)	-0.009 (0.020)	1,965	Electronics	0.015* (0.007)	1,965
South East	0.003 (0.017)	1,965	Environment	0.005 (0.005)	1,965
South West	0.003 (0.018)	1,965	Food & drink	0.003 (0.012)	1,965
West Midlands	0.002 (0.020)	1,965	Financial services	0.002 (0.003)	1,965
Yorkshire & Humberside	-0.005 (0.016)	1,965	Fire & security	-0.019 (0.013)	1,965
Other	0.018 (0.014)	1,965	Jewellery/tableware	0.003 (0.006)	1,965
Capabilities			Healthcare	-0.013 (0.013)	1,965
1: Managing people	0.039 (0.047)	1,905	Household goods	0.000 (0.008)	1,965
2: Developing a business plan	0.049 (0.053)	1,954	Leisure & tourism	0.008 (0.007)	1,965
3: Accessing the UK market	0.020 (0.057)	1,852	Mechanical/electric	0.001 (0.008)	1,965
4: Accessing other markets	0.082 (0.077)	1,555	Oil & gas	-0.005 (0.005)	1,965
5: Developing new products	0.081 (0.058)	1,924	Ports & logistics	-0.001 (0.003)	1,965
6: Accessing external finance	-0.025 (0.071)	1,722	Computer services	-0.009 (0.013)	1,965
7: Recruiting new employees	0.027 (0.057)	1,835	Sports & leisure	-0.011 (0.009)	1,965
8: Using IT to grow	0.041 (0.067)	1,951	Others	-0.017 (0.024)	1,965

Table A.4. Balance of treatment and control groups at baseline for firms interviewed in the 24-month survey

Note: Each row corresponds to an OLS regression of voucher allocation on a firm characteristic. The covariates correspond to month of application, cohort and the full set of interactions between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

	ID	BR	12-month survey 24		24-mont	4-month survey	
	(1)	(2)	(1)	(2)	(1)	(2)	
Allocation of voucher	0.344** (0.005)	0.344** (0.004)	0.438** (0.008)	0.434** (0.008)	0.514** (0.013)	0.512** (0.014)	
Basic controls & Full set of interactions	No	Yes	No	Yes	No	Yes	
Obs.	13,444	13,444	5,579	5,579	1,965	1,965	

Table A.5. First stage of IV estimation for firms in survey data and IDBR

Note: OLS regression for use of the voucher on allocation of voucher. Basic controls include month of application and cohort. The full set of interactions corresponds to all combinations between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. Robust standard errors in parentheses. \*\* p<0.01, \* p<0.05.

Table A.6.	OLS and IV	<sup>7</sup> estimates	for	survival	

	OLS	OLS	OLS	IV	IV	IV	
	(1)	(2)	(3)	(1)	(2)	(3)	
		S	urvival afte	er 12 months			
Allocation of voucher	-0.003 (0.006)	-0.003 (0.006)	-0.003 (0.006)	N/A	N/A	N/A	
Take up of the voucher	N/A	N/A	N/A	-0.010 (0.017)	-0.008 (0.017)	-0.009 (0.017)	
Obs.	$13,\!274$	$13,\!274$	13,274	13,274	13,274	$13,\!274$	
		S	urvival afte	er 24 monti	hs		
Allocation of voucher	-0.008 (0.007)	-0.008 (0.007)	-0.008 (0.007)	N/A	N/A	N/A	
Take up of the voucher	N/A	N/A	N/A	-0.023 (0.020)	-0.022 (0.020)	-0.023 (0.020)	
Obs.	$13,\!274$	$13,\!274$	$13,\!274$	$13,\!274$	$13,\!274$	$13,\!274$	
		S	urvival afte	er 36 monti	hs		
Allocation of voucher	-0.001 (0.008)	-0.001 (0.008)	-0.001 (0.008)	N/A	N/A	N/A	
Take up of the voucher	N/A	N/A	N/A	-0.003 (0.023)	-0.002 (0.023)	-0.002 (0.023)	
Obs.	$13,\!274$	$13,\!274$	$13,\!274$	$13,\!274$	13,274	$13,\!274$	
Basic controls	No	Yes	Yes	No	Yes	Yes	
Type of diagnostic & theme of advice	No	Yes	N/A	No	Yes	N/A	
Full set of interactions	No	No	Yes	No	No	Yes	

Note: OLS and IV regressions for survival status since baseline. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05. Basic controls include month of application and cohort. The full set of interactions corresponds to all combinations between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. N/A: not applicable.

	OLS	OLS	IV	IV
	(1)	(2)	(1)	(2)
		Turnover aft	ter 12 months	-
Allocation of voucher	0.026 (0.014)	0.026 (0.014)	N/A	N/A
Take up of the voucher	N/A	N/A	0.072 (0.039)	0.071 (0.039)
Obs.	8,827	8,827	8,827	8,827
		Turnover aft	ter 24 months	
Allocation of voucher	0.012 (0.022)	0.011 (0.022)	N/A	N/A
Take up of the voucher	N/A	N/A	0.032 (0.061)	0.029 (0.061)
Obs.	8,827	8,827	8,827	8,827
		Employment &	after 12 months	
Allocation of voucher	-0.007 (0.007)	-0.008 (0.007)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.020 (0.021)	-0.022 (0.021)
Obs.	5,859	5,859	5,859	5,859
		Employment &	after 24 months	
Allocation of voucher	-0.005 (0.012)	-0.005 (0.012)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.014 (0.033)	-0.015 (0.033)
Obs.	5,859	5,859	5,859	5,859
		Employment &	after 36 months	
Allocation of voucher	0.002 (0.015)	0.001 (0.015)	N/A	N/A
Take up of the voucher	N/A	N/A	0.005 (0.043)	0.004 (0.044)
Obs.	5,859	5,859	5,859	5,859
Basic controls & full set of interactions	No	Yes	No	Yes

Table A.7. Robustness check 1: Using the same sample of firms across time for each outcome

interactionsNotNote: OLS and IV regressions for percentage change since baseline. Basic controls include month of<br/>application and cohort. The full set of interactions corresponds to all combinations between type of<br/>diagnostic, theme of advice and a dummy indicating if the firm applied in last month. N/A: not applicable.<br/>Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.</td>

	OLS	OLS	IV	IV
	(1)	(2)	(1)	(2)
		Turnover aft	ter 12 months	
Allocation of voucher	0.040* (0.017)	0.039* (0.017)	N/A	N/A
Take up of the voucher	N/A	N/A	0.108* (0.047)	0.106* (0.047)
Obs.	6,442	6,442	6,442	6,442
		Turnover aft	ter 24 months	
Allocation of voucher	0.019 (0.028)	0.017 (0.028)	N/A	N/A
Take up of the voucher	N/A	N/A	0.051 (0.074)	0.046 (0.074)
Obs.	6,040	6,040	6,040	6,040
		Employment a	ofter 12 months	
Allocation of voucher	-0.013 (0.008)	-0.013 (0.008)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.036 (0.024)	-0.037 (0.024)
Obs.	5,020	5,020	5,020	5,020
		Employment a	ofter 24 months	
Allocation of voucher	-0.012 (0.013)	-0.012 (0.013)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.034 (0.038)	-0.034 (0.038)
Obs.	4,550	4,550	4,550	4,550
		Employment a	ofter 36 months	
Allocation of voucher	0.002 (0.018)	0.002 (0.018)	N/A	N/A
Take up of the voucher	N/A	N/A	0.006 (0.050)	0.007 (0.051)
Obs.	4,175	4,175	4,175	4,175
Basic controls & full set of interactions	No	Yes	No	Yes

Table A.8. Robustness check 2: Excluding firms that enrolled in the last month of the application period

 interactions
 No
 res
 No
 res

 Note: OLS and IV regressions for percentage change since baseline. Basic controls include month of application and cohort. The interaction term corresponds to a double interaction between type of diagnostic and theme of advice. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.</td>

	OLS	OLS	IV	IV
	(1)	(2)	(1)	(2)
		Turnover aft	ter 12 months	-
Allocation of voucher	0.038* (0.016)	0.039* (0.016)	N/A	N/A
Take up of the voucher	N/A	N/A	0.098* (0.041)	0.102* (0.041)
Obs.	7,825	7,825	7,825	7,825
		Turnover aft	ter 24 months	
Allocation of voucher	0.017 (0.025)	0.018 (0.025)	N/A	N/A
Take up of the voucher	N/A	N/A	0.045 (0.064)	0.047 (0.064)
Obs.	7,339	7,339	7,339	7,339
		Employment a	after 12 months	
Allocation of voucher	-0.009 (0.008)	-0.009 (0.008)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.024 (0.021)	-0.025 (0.021)
Obs.	6,065	6,065	6,065	6,065
		Employment a	after 24 months	
Allocation of voucher	-0.013 (0.012)	-0.013 (0.012)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.036 (0.033)	-0.036 (0.033)
Obs.	5,485	5,485	5,485	5,485
		Employment a	after 36 months	
Allocation of voucher	-0.006 (0.016)	-0.006 (0.016)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.017 (0.045)	-0.017 (0.044)
Obs.	5,066	5,066	5,066	5,066
Basic controls & full set of	No	Yes	No	Yes

Table A.9. Robustness check 3: Excluding firms that did not select the suggested theme of advice

interactionsINOI esINOI esNote: OLS and IV regressions for percentage change since baseline. Basic controls include month of<br/>application and cohort. The full set of interactions corresponds to all combinations between type of<br/>diagnostic, theme of advice and a dummy indicating if the firm applied in last month. N/A: not applicable.<br/>Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.</td>

	OLS	OLS	IV	IV
	(1)	(2)	(1)	(2)
		Turnover aft	ter 12 months	-
Allocation of voucher	0.030* (0.014)	0.030* (0.014)	N/A	N/A
Take up of the voucher	N/A	N/A	0.090* (0.043)	0.089* (0.043)
Obs.	9,452	9,452	9,452	9,452
		Turnover aft	ter 24 months	
Allocation of voucher	0.010 (0.022)	0.009 (0.022)	N/A	N/A
Take up of the voucher	N/A	N/A	0.029 (0.066)	0.026 (0.066)
Obs.	8,855	8,855	8,855	8,855
		Employment &	after 12 months	
Allocation of voucher	-0.008 (0.007)	-0.009 (0.007)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.027 (0.022)	-0.028 (0.022)
Obs.	7,326	7,326	7,326	7,326
		Employment a	after 24 months	
Allocation of voucher	-0.007 (0.011)	-0.008 (0.011)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.023 (0.034)	-0.024 (0.034)
Obs.	6,606	6,606	6,606	6,606
		Employment a	after 36 months	
Allocation of voucher	0.000 (0.015)	-0.000 (0.015)	N/A	N/A
Take up of the voucher	N/A	N/A	0.000 (0.046)	-0.001 (0.046)
Obs.	6,084	6,084	6,084	6,084
Basic controls & full set of interactions	No	Yes	No	Yes

Table A.10. Robustness check 4: Treating non-recipients that paid for GVP advice as participants

interactionsINOI esINOI esNote: OLS and IV regressions for percentage change since baseline. Basic controls include month of<br/>application and cohort. The full set of interactions corresponds to all combinations between type of<br/>diagnostic, theme of advice and a dummy indicating if the firm applied in last month. N/A: not applicable.<br/>Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.</td>

	OLS	OLS	IV	IV
	(1)	(2)	(1)	(2)
		Turnover aft	er 12 months	
Allocation of voucher	0.030* (0.014)	0.030* (0.014)	N/A	N/A
Take up of the voucher	N/A	N/A	0.092* (0.044)	0.091* (0.044)
Obs.	9,452	9,452	9,452	9,452
		Turnover aft	er 24 months	
Allocation of voucher	0.010 (0.022)	0.009 (0.022)	N/A	N/A
Take up of the voucher	N/A	N/A	0.031 (0.069)	0.028 (0.069)
Obs.	8,855	8,855	8,855	8,855
		Employment a	fter 12 months	
Allocation of voucher	-0.008 (0.007)	-0.009 (0.007)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.028 (0.023)	-0.029 (0.023)
Obs.	7,326	7,326	7,326	7,326
		Employment a	fter 24 months	
Allocation of voucher	-0.007 (0.011)	-0.008 (0.011)	N/A	N/A
Take up of the voucher	N/A	N/A	-0.025 (0.036)	-0.025 (0.036)
Obs.	6,606	6,606	6,606	6,606
		Employment a	fter 36 months	
Allocation of voucher	0.000 (0.015)	-0.000 (0.015)	N/A	N/A
Take up of the voucher	N/A	N/A	0.000 (0.049)	-0.001 (0.049)
Obs.	6,084	6,084	6,084	6,084
Basic controls & full set of interactions	No	Yes	No	Yes

Table A.11. Robustness check 5: Extending the definition of treatment to include firms that paid for any type of non-GVP advice

Note: OLS and IV regressions for percentage change since baseline. Basic controls include month of application and cohort. The full set of interactions corresponds to all combinations between type of diagnostic, theme of advice and a dummy indicating if the firm applied in last month. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

	Pers	sonal	Onl	ine		
	OLS	IV	OLS	IV		
	(1)	(2)	(1)	(2)		
		Turnover aft	er 12 months			
Allocation of voucher	0.033 (0.021)	N/A	0.025 (0.020)	N/A		
Take up of the voucher	N/A	0.080 (0.050)	N/A	0.083 (0.064)		
Obs.	4,759	4,759	4,693	4,693		
		Turnover aft	er 24 months			
Allocation of voucher	0.000 (0.032)	N/A	0.012 (0.030)	N/A		
Take up of the voucher	N/A	0.000 (0.077)	N/A	0.039 (0.097)		
Obs.	4,453	4,453	4,402	4,402		
		Employment a	after 12 months			
Allocation of voucher	-0.018 (0.010)	N/A	0.000 (0.010)	N/A		
Take up of the voucher	N/A	-0.046 (0.025)	N/A	0.001 (0.034)		
Obs.	3,702	3,702	3,624	3,624		
		Employment a	fter 24 months			
Allocation of voucher	-0.024 (0.016)	N/A	0.011 (0.015)	N/A		
Take up of the voucher	N/A	-0.061 (0.039)	N/A	0.038 (0.052)		
Obs.	3,360	3,360	3,246	3,246		
		Employment a	fter 36 months			
Allocation of voucher	-0.006 (0.022)	N/A	$0.009 \\ (0.021)$	N/A		
Take up of the voucher	N/A	-0.016 (0.054)	N/A	0.030 (0.070)		
Obs.	3,094	3,094	2,990	2,990		

Table A.12. OLS and IV estimates by type of diagnostic for firms in the IDBR

Note: OLS and IV regressions for percentage change since baseline using our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

	Sales & n	narketing	Raising	finance	Leadership &	management	Workforce d	evelopment	Digital t	gital technologies	
	OLS	IV	OLS	IV	OLS	IV	OLS	IV	OLS	IV	
	(1)	(2)	(1)	(2)	(1)	(2)	(1)	(2)	(1)	(2)	
			<u>.</u>	Turnov	ver after 12 mon	ths					
Allocation of voucher	0.068** (0.021)	N/A	-0.052 (0.049)	N/A	0.024 (0.031)	N/A	-0.005 (0.050)	N/A	0.005 (0.028)	N/A	
Take up of the voucher	N/A	$0.188^{**}$ (0.059)	N/A	-0.132 (0.125)	N/A	0.074 (0.093)	N/A	-0.029 (0.286)	N/A	0.013 (0.071)	
Obs.	4,354	4,354	1,220	1,220	1,388	1,388	495	495	1,995	1,995	
				Turnov	ver after 24 mon	ths			_		
Allocation of voucher	0.073* (0.032)	N/A	-0.155 (0.084)	N/A	0.003 (0.044)	N/A	-0.007 (0.087)	N/A	-0.023 (0.046)	N/A	
Take up of the voucher	N/A	0.198* (0.089)	N/A	-0.390 (0.210)	N/A	0.010 (0.129)	N/A	-0.042 (0.491)	N/A	-0.058 (0.114)	
Obs.	4,078	4,078	1,122	1,122	1,319	1,319	463	463	1,873	1,873	
				Employn	nent after 12 me	onths					
Allocation of voucher	-0.008 (0.010)	N/A	-0.017 (0.021)	N/A	0.003 (0.017)	N/A	-0.050 (0.033)	N/A	-0.002 (0.015)	N/A	
Take up of the voucher	N/A	-0.025 (0.031)	N/A	-0.043 (0.054)	N/A	0.010 (0.052)	N/A	-0.269 (0.181)	N/A	-0.004 (0.039)	
Obs.	3,392	3,392	976	976	1,103	1,103	389	389	1,466	1,466	
				Employn	nent after 24 m	onths			_		
Allocation of voucher	-0.004 (0.016)	N/A	-0.020 (0.032)	N/A	-0.004 (0.029)	N/A	-0.057 (0.052)	N/A	0.005 (0.023)	N/A	
Take up of the voucher	N/A	-0.013 (0.047)	N/A	-0.049 (0.079)	N/A	-0.013 (0.088)	N/A	-0.322 (0.299)	N/A	0.013 (0.061)	
Obs.	3,083	3,083	866	866	993	993	355	355	1,309	1,309	
				Employn	nent after 36 m	onths			_		
Allocation of voucher	0.006 (0.022)	N/A	-0.045 (0.046)	N/A	-0.039 (0.040)	N/A	0.020 (0.065)	N/A	0.042 (0.031)	N/A	
Take up of the voucher	N/A	0.019 (0.064)	N/A	-0.112 (0.113)	N/A	-0.118 (0.121)	N/A	0.110 (0.361)	N/A	0.107 (0.079)	
Obs.	2,849	2,849	768	768	932	932	328	328	1,207	1,207	

## Table A.13. OLS and IV estimates for firms in the IDBR by theme of advice

Note: Regressions for percentage change since baseline using our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. N/A: not applicable. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.



Figure A.1. Distribution of the propensity score for treated and control firms Turnover after 12 months Turnover after 24 months

Note: For each outcome and period, the graph shows the distribution of the estimated propensity score using our preferred probit specification for the group of GVP compliers and the matched control group.

	After 12	months	After 24	months
	Coeff.	Obs.	Coeff.	Obs.
Veens the diam	-0.207	5 205	-0.044	5 159
Tears trading	(0.193)	0,590	(0.198)	0,100
Turnovor in 2015	-12.307	5 395	-49.448	5 159
	(54.178)	0,000	(53.933)	0,100
Turnover in 2014	-8.308	5.395	-39.343	5.153
	(46.952)	0,000	(46.221)	0,100
Employment in 2015	0.355	5,395	0.176	5,153
T	(0.225)	,	(0.223)	,
Location				
East of England	0.001	5,395	-0.008	5,153
0	(0.008)	·	(0.009)	
East of Midlands	0.008	5,395	0.006	5,153
	(0.008)		(0.008)	
Greater London	(0.008)	5,395	(0.005)	5,153
	0.010		0.010	
Home counties	(0.000)	5,395	(0.010)	5,153
	-0.001		0.006	
North East	(0.006)	5,395	(0.006)	5,153
NT 11 TTT 1	-0.001		-0.002	
North West	(0.011)	5,395	(0.011)	5,153
Q. d. E	-0.008	<b>F</b> 90 <b>F</b>	-0.003	F 150
South East	(0.008)	5,395	(0.008)	5,153
Carth West	-0.013	F 90F	-0.005	F 159
South west	(0.010)	5,395	(0.010)	0,100
West Midlands	0.008	5 205	0.001	5 159
west mulanus	(0.010)	5,555	(0.011)	0,100
Vorkshiro & Humborsido	-0.007	5 395	-0.008	5 1 5 3
Torkshile & Humberside	(0.010)	0,000	(0.010)	0,100
Other	-0.001	5 395	-0.002	5 153
	(0.004)	0,000	(0.004)	0,100
Sector				
Agriculturo forestry & fishing	0.000	5 395	-0.001	5 1 5 3
Agriculture, lorestry & lishing	(0.003)	0,000	(0.004)	0,100
Manufacturing	-0.006	5 395	-0.014	5 153
hianalaobaling	(0.009)	0,000	(0.009)	0,100
Construction	0.007	5.395	0.003	5.153
	(0.007)	-,	(0.007)	0,200
Wholesale & retail	-0.012	5,395	-0.031*	5,153
	(0.013)	·	(0.013)	
Transportation & storage	(0.003)	5,395	(0.001)	5,153
	(0.003)		-0.001	
Accommodation & food services	(0.001)	5,395	(0.001)	5,153
	0.024*		(0.005)	
Information & communication	(0.024)	5,395	(0.028)	5,153
	-0.000		-0.002	
Real estate	(0.005)	5,395	(0.002)	5,153
	0.005		0.017	
Professional, scientific & technical	(0.014)	5,395	(0.014)	5,153
	-0.007		-0.003	<b>_</b>
Administrative & support services	(0.010)	5,395	(0.010)	5,153
	-0.002		0.002	<b>.</b>
Education	(0.005)	5,395	(0.005)	5,153
	-0.013		0.002	<b>F</b> 1 <b>F</b> 0
Otner	(0.008)	5,395	(0.008)	5,153

Table A.14. Balance of treatment and control groups after controlling for the propensity score for turnover

Note: Each row corresponds to an OLS regression of treatment status on a firm characteristic, controlling for the propensity score. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

	After 12	months	After 24	months	After 36	months
	Coeff.	Obs.	Coeff.	Obs.	Coeff.	Obs.
Years trading	-0.086 (0.195)	5,455	-0.067 (0.204)	5,061	-0.163 (0.211)	4,688
Turnover in 2015	-54.690 (47.623)	5,455	-31.829 (63.659)	5,061	-72.497 (99.004)	4,688
Turnover in 2014	-59.197 (43.817)	5,455	-4.092 (41.708)	5,061	-45.595 (60.023)	4,688
Employment in 2015	0.019 (0.187)	5,455	0.424* (0.201)	5,061	0.434* (0.211)	4,688
Location						
East of England	-0.003 (0.008)	5,455	0.006 (0.008)	5,061	0.001 (0.009)	4,688
East of Midlands	0.001 (0.008)	5,455	-0.015 (0.009)	5,061	0.008 (0.008)	4,688
Greater London	0.010 (0.010)	5,455	0.008 (0.010)	5,061	0.014 (0.011)	4,688
Home counties	-0.004 (0.007)	5,455	0.010 (0.007)	5,061	0.007 (0.007)	4,688
North East	-0.005 (0.006)	5,455	-0.004 (0.006)	5,061	0.003 (0.007)	4,688
North West	0.008 (0.011)	5,455	0.000 (0.011)	5,061	-0.005 (0.012)	4,688
South East	-0.002 (0.008)	5,455	-0.006 (0.008)	5,061	0.003 (0.009)	4,688
South West	-0.002 (0.010)	5,455	-0.006 (0.010)	5,061	-0.015 (0.011)	4,688
West Midlands	0.003 (0.011)	5,455	-0.002 (0.011)	5,061	-0.016 (0.012)	4,688
Yorkshire & Humberside	-0.004 (0.010)	5,455	0.007 (0.010)	5,061	0.000 (0.011)	4,688
Other	-0.001 (0.004)	5,455	0.002 (0.003)	5,061	-0.001 (0.004)	4,688
Sector						
Agriculture, forestry & fishing	-0.000 (0.004)	5,455	-0.004 (0.004)	5,061	-0.002 (0.004)	4,688
Manufacturing	-0.013 (0.009)	5,455	-0.012 (0.009)	5,061	0.009 (0.009)	4,688
Construction	0.005 (0.007)	5,455	0.001 (0.007)	5,061	0.000 (0.007)	4,688
Wholesale & retail	-0.002 (0.012)	5,455	-0.006 (0.013)	5,061	-0.003 (0.014)	4,688
Transportation & storage	-0.003 (0.004)	5,455	0.003 (0.003)	5,061	-0.001 (0.004)	4,688
Accommodation & food services	-0.002 (0.005)	5,455	0.002 (0.004)	5,061	0.002 (0.004)	4,688
Information & communication	0.019 (0.010)	5,455	0.017 (0.010)	5,061	0.020 (0.010)	4,688
Real estate	0.000 (0.005)	5,455	-0.001 (0.005)	5,061	0.004 (0.005)	4,688
Professional, scientific & technical	0.021 (0.014)	5,455	0.005 (0.014)	5,061	-0.006 (0.015)	4,688
Administrative & support services	-0.008 (0.010)	5,455	-0.005 (0.010)	5,061	-0.003 (0.010)	4,688
Education	-0.003 (0.005)	5,455	0.000 (0.005)	5,061	-0.000 (0.005)	4,688
Other	-0.014 (0.008)	5,455	(0.001)	5,061	-0.021* (0.009)	4,688

Table A.15. Balance of treatment and control groups after controlling for the propensity score for employment

Note: Each row corresponds to an OLS regression of treatment status on a firm characteristic, controlling for the propensity score. Robust standard errors in parentheses; \*\* p<0.01, \* p<0.05.

Outcome	No.	Observables used for the matching process	ATT	Avg. diff. in the PS	Significant differences at baseline after controlling for the PS
Turnover after 12 months	(1)	Turnover & employment in 2015	0.073** (0.020)	0.0037* (0.0017)	Age, 9 regions and 9 sectors
	(2)	Turnover & employment in 2015, age and aged squared	0.076** (0.019)	-0.0002 (0.0019)	8 regions and 7 sectors
	(3)	Turnover & employment in 2015, age, aged squared and region	0.059** (0.020)	-0.0007 (0.0033)	1 region and 8 sectors
	(4)	Turnover & employment in 2015, age, aged squared, region and sector	0.064** (0.021)	-0.0008 (0.0037)	1 region and 1 sector
	(5)	Turnover & employment in 2015, age, aged squared, region, sector, turnover & employment in 2014, and interaction term for London and the biggest sectors	0.063** (0.020)	-0.0014 (0.0041)	1 sector
Turnover after 24 months	(1)	Turnover & employment in 2015	0.135** (0.028)	0.0041* (0.0021)	Age, 9 regions and 9 sectors
	(2)	Turnover & employment in 2015, age and aged squared	0.135** (0.028)	-0.0004 (0.0024)	10 regions and 9 sectors
	(3)	Turnover & employment in 2015, age, aged squared and region	0.112** (0.031)	-0.0005 (0.0038)	9 sectors
	(4)	Turnover & employment in 2015, age, aged squared, region and sector	0.126** (0.029)	-0.0010 (0.0043)	2 sectors
	(5)	Turnover & employment in 2015, age, aged squared, region, sector, turnover & employment in 2014, and interaction term for London and the biggest sectors, and interaction term for London and the biggest sectors	0.125** (0.032)	-0.0012 (0.0047)	None

Table A.16. PSM specification search for turnover

Note: Bootstrapped standard errors in parentheses; \*\* p<0.01, \* p<0.05.

Outcome No. Observables used for the matching process		ATT	Avg. diff. in the PS	Significant differences at baseline after controlling for the PS	
	(1)	Turnover & employment in 2015	0.049** (0.008)	0.0043* (0.0017)	Age, 10 regions and 8 sectors
	(2)	Turnover & employment in 2015, age and aged squared	0.036** (0.007)	-0.0001 (0.0018)	9 regions and 7 sectors
Employment	(3)	Turnover & employment in 2015, age, aged squared and region	0.039** (0.008)	-0.0004 (0.0031)	Turnover in 2014 and 2015 and 6 sectors
months	(4)	Turnover & employment in 2015, age, aged squared, region and sector	0.043** (0.008)	-0.0016 (0.0035)	Employment in 2015 and 2 sectors
	(5)	Turnover & employment in 2015, age, aged squared, region, sector, turnover & employment in 2014, and interaction term for London and the biggest sectors	0.031** (0.008)	-0.0008 (0.0041)	None
	(1)	Turnover & employment in 2015	0.074** (0.012)	0.0016* (0.0007)	Age, employment in 2015, 7 regions and 8 sectors
	(2)	Turnover & employment in 2015, age and aged squared	0.065** (0.013)	-0.0002 (0.0010)	8 regions and 6 sectors
Employment after 24	(3)	Turnover & employment in 2015, age, aged squared and region	0.072** (0.013)	-0.0003 (0.0032)	Employment in 2015, 1 region and 6 sectors
months	(4)	Turnover & employment in 2015, age, aged squared, region and sector	0.072** (0.013)	-0.0010 (0.0036)	Age, employment in 2015, 1 region and 3 sectors
	(5)	Turnover & employment in 2015, age, aged squared, region, sector, turnover & employment in 2014, and interaction term for London and the biggest sectors	0.077** (0.012)	-0.0010 (0.0042)	Employment in 2015
	(1)	Turnover & employment in 2015	0.124** (0.016)	0.0011 (0.0009)	Age, 7 regions and 8 sectors
	(2)	Turnover & employment in 2015, age and aged squared	0.108** (0.016)	-0.0002 (0.0012)	Turnover in 2014 and 2015, employment in 2015, and 7 sectors
Employment	(3)	Turnover & employment in 2015, age, aged squared and region	0.114** (0.017)	-0.0007 (0.0035)	Turnover in 2014 and 2015, employment in 2015, and 7 sectors
months	(4)	Turnover & employment in 2015, age, aged squared, region and sector	0.101** (0.017)	-0.0011 (0.0041)	Turnover in 2014 and 2015, employment in 2015, and 1 sector
	(5)	Turnover & employment in 2015, age, aged squared, region, sector, turnover & employment in 2014, and interaction term for London and the biggest sectors	0.100** (0.018)	-0.0008 (0.0047)	Employment in 2015, and 1 sector

Table A.17. PSM spe	ecification searc	h for em	ployment
---------------------	-------------------	----------	----------

Note: Bootstrapped standard errors in parentheses; \*\* p<0.01, \* p<0.05.

#### A.3 Heterogeneity by size and age of the firm

The Growth Vouchers Programme was designed with the expectation that small and young firms would benefit the most from receiving diagnostic support and subsidised business advice.<sup>50</sup> The existing literature shows mixed evidence regarding age and size cut-offs above and below which programmes are more likely to benefit participant firms. See for example Criscuolo, et al. (2012), who find that the regional selective assistance (RSA) programme in the UK was only effective for small firms with fewer than 150 employees. In contrast, Wren and Storey (2002) find that marketing advice in the UK had the highest positive impact for 10-80 employee firms, with no impact on smaller firms and a negative impact on survival for large SMEs. Additional evidence suggests that more established firms were more likely to benefit from training on finance and accounting (Anderson, Chandy and Zia, 2018) and that there may be decreasing returns to business advice with firm size (Bruhn et al., 2018). In line with the broader literature, our results from GVP are also inconclusive when obtaining estimates by the size and age of firms at the time of application or baseline (using IDBR data).

Figure A.2 presents the IV results for the impact of GVP on turnover and employment for firms under a specific age threshold at baseline. The figure shows each point estimate and the associated 95% confidence interval. ATT estimates indicate GVP had a positive impact on turnover after 12 months for two firm age groups: among all firms aged 3 years or less, and among all firms aged 4 years or less. This effect is not significant when considering other age cut-offs, nor after 24 months. In addition, we find no effect of GVP on employment for any age sub-group. Overall, there is no clear pattern in terms of the effect of GVP on turnover and employment by firm age.

Figure A.3 presents the IV results for firms of different size or number of employees. GVP had a positive and significant impact on turnover after 12 months for all size cutoffs considered (except for firms with 3 employees or less due to large standard errors). However, IV turnover estimates are not significant after 24 months. Further, the results indicate no significant effects on employment in any period. As with age, we do not find a clear pattern for the effect of GVP on firm performance by size of the firm.

<sup>&</sup>lt;sup>50</sup> This is reflected in the eligibility criteria of GVP. See BIS (2013) as well as page 9 of the full trial protocol, available at https://www.gov.uk/government/publications/growth-vouchers-programme-trial-protocol.



Figure A.2. Heterogeneity by firm age at baseline Turnover after 12 months

Note: Each coefficient corresponds to an IV estimate considering firms below the indicated age threshold. IV estimates for percentage changes since baseline using our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. The vertical lines depict 95% confidence intervals.



#### Figure A.3. Heterogeneity by firm size Turnover after 12 months

Note: Each coefficient corresponds to an IV estimate considering firms below the indicated size threshold. IV estimates for percentage changes since baseline using our preferred specification, which controls for cohort and month of application and includes the full set of interactions between type of diagnostic, theme of advice and dummy for last month of application. The vertical lines depict 95% confidence intervals.

#### A.4 Details on the identification strategies

This section presents the instrumental variables identification strategy and the propensity score matching approach.

#### Instrumental variables

To get an estimate of the effect of treatment on the treated we rely on the fact that the random assignment of a voucher by construction influences the probability of using the voucher (as vouchers are non-transferable). To do this we employ an instrumental variables (IV) approach using the variation in take up of the voucher induced by the allocation of the voucher to estimate the effect of treatment on the treated (ATT). If eligible, the assignment to a group can be defined by:

Z = 1 for treatment and Z = 0 for control

while the actual treatment decision  $D_z$  corresponds to

 $D_1 = 1$  Treatment status if used the voucher

 $D_0 = 0$  Treatment status if assigned to control group

Overall, we have 4 groups according to the allocation and use of the voucher

- 1) Compliers:  $D_1 > D_0$  ( $D_0 = 0$  and  $D_1 = 1$ )
- 2) Always-takers:  $D_1 = D_0 = 1$
- 3) Never-takers:  $D_1 = D_0 = 0$
- 4) Defiers:  $D_1 < D_0 (D_0 = 1 \text{ and } D_1 = 0)$

To identify the impact of the programme on the treated we estimate:

$$E[Y_1 - Y_0 | D_1 > D_0] = \frac{E[Y|Z=1] - E[Y|Z=0]}{E[D|Z=1] - E[D|Z=0]} = \frac{cov(Y,Z)}{cov(D,Z)}$$

where E  $[Y_1 - Y_0 | D_1 > D_0]$  corresponds to the expected change in outcome Y between the treatment and control groups, conditioned on complying with the treatment. In this way, the IV estimator corresponds to the average treatment effect for the subpopulation of compliers. This provides an estimate of the local average treatment effect (LATE) or average treatment effect on the treated (ATT) for the firms whose participation status changes due to their selection into the treatment group. The identification assumptions are:

- 1) Independence:  $(Y_0, Y_1, D_0, D_1) \coprod \mathbb{Z}$
- 2) First stage: 0 < P(Z = 1) < 1 and  $P(D_1 = 1) \neq P(D_0 = 1)$
- 3) Monotonicity:  $D_1 \ge D_0$

For the GVP, we have one-side non-compliance  $(D_1 = 0)$ , and thus

$$E [Y_1 | D_1 > D_0] = E [Y_1 | D_1 = 1] = E [Y_1 | Z_1 = 1, D_1 = 1] = E [Y_1 | D = 1]$$

Similarly,

$$E[Y_0 | D_1 > D_0] = E[Y_0 | D = 1]$$

such that the local average treatment effect for compliers corresponds to the average treatment effect on the treated (ATT) in the following way

LATE = E 
$$[Y_1 - Y_0 | D_1 > D_0]$$
 = E  $[Y_1 - Y_0 | D = 1]$  = ATT

Therefore, the IV strategy estimates the effects of receiving business support for compliers, which corresponds to the ATT, instead of estimating the effect of being offered subsidised advice (the ITT effect). The first stage of the IV estimation is:

$$P_i = \omega + \delta V_i + \gamma X_i + u_i$$

where  $P_i$  measures whether business *i* used the voucher received as part of the GVP,  $V_i$  corresponds to the allocation of voucher for each business and  $X_i$  is a vector of baseline characteristics or covariates. The second stage regression for any outcome Y is

$$Y_i = \alpha + \beta \hat{P}_i + \rho X_i + \varepsilon_i$$

where  $\hat{P}_i$  is the predicted likelihood for firm *i* to participate in the GVP given that it was randomly allocated a voucher for business support. As discussed above, the coefficient of interest is  $\beta$  which provides an estimate of the ATT.

#### Propensity score matching

PSM imputes the missing potential outcome of each treated unit using the observed outcome from a similar untreated unit, where similarity is defined in terms of a set of observable covariates, such that:

$$ATT = \frac{1}{N} \sum_{D_i=1} (Y_i - Y_{j(i)})$$

Where *N* corresponds to the total number of treated observations  $D_i$ ,  $Y_i$  to the outcome of the treated observation,  $Y_{j(i)}$  is the outcome of an untreated observation such that  $X_{j(i)}$  is the closest value for covariate  $X_i$  among the untreated observations. If we consider the average of the M closest matches, then

$$ATT = \frac{1}{N} \sum_{D_i=1} \left\{ Y_i - \left( \frac{1}{M} \sum_{m=1}^M Y_{jm(i)} \right) \right\}$$

The discrepancy between treated and untreated observations tends to increase with the number of covariates  $X_i$  considered, since it becomes difficult to find units that are similar on all of these dimensions. Thus, a common solution is to estimate and use a propensity score (PS), such that this score is identified as the selection probability conditional on the confounding or observable variables. The exogeneity assumption corresponds to:

$$(Y_1, Y_0) \perp D \mid ps(x)$$

That is, the selection on observables conditioning on the propensity score is enough to have independence between the treatment indicator D and the potential outcome of interest Y. A second condition requires having common support or probability of assignment bounded by 0 < ps(x) < 1. The allocation mechanism can be interpreted as if, within subpopulations of units with the same value for the score, a random assignment was carried out. If both assumptions hold, then:

$$ATT = E[Y_1 - Y_0 | D = 1] = \frac{1}{P(D = 1)} E[Y \frac{D - p(X)}{1 - p(X)}]$$

#### A.5 Distribution of firms between tradable and non-tradable sectors

In this section we consider how many GVP firms belong to tradable and non-tradable sectors. To do this, we follow the methodology developed by Jensen and Kletzer (2006), which uses the geographical concentration of service activities (within the US) to identify whether activities are traded domestically or not. They assume that geographically concentrated activities are more tradable – because they can serve dispersed customers from a smaller number of locations. They define three degrees of tradability: Gini class 1 (least geographically concentrated, when the Gini index < 0.1); Gini class 2 (when the Gini index is between 0.1 and 0.3); and Gini class 3 (most geographically concentrated, when the Gini index is >0.3).

Group and measure			Services sector					
		Manufacturing sector	Tradable (Gini 3)	Medium tradable (Gini 2)	Non-tradable (Gini 1)	Other sector	Not classified	Total
All GVP firms in IDBR	Number	1,281	1,925	6,038	3,190	955	55	13,444
	Share by sector	9.5%	14.3%	44.9%	23.7%	7.1%	0.4%	100%
GVP compliers	Number	362	460	1,615	800	229	13	3,479
	Share by sector	10.4%	13.2%	46.4%	23.0%	6.6%	0.4%	100%

Table A.18. Breakdown of compliers by sectoral classification

Note: Non-tradable services also include the public sector and construction services. Gini 1, Gini 2 and Gini 3 are derived following Jensen and Kletzer (2006). 'Other' are sectors that are not classified by degree of tradability because they were either excluded in Faggio & Overman (2014) (e.g., electricity and gas, transport, or telecommunications) or they were not mapped into a degree of tradability in the correspondence process (e.g., some administrative support activities).

To use this classification for UK sectors, we proceed as follows. We first use the mapping by Faggio & Overman (2014) from the US 6-digit codes to the UK 4-digit SIC03 codes based on an industry correspondence table from the US Census Bureau. The correspondence table maps 2002 NAICS (6-digit code) to NACE Rev. 1.1 (4-digit code) which is almost equivalent to the UK SIC03 classification. Together with the 4-digit code, the correspondence table also provides a brief description of the industry. Faggio & Overman (2014) combine information on the industry code and description to match the NAICS and the UK SIC classifications. In a second step, we map the UK 5-digit SIC03 codes to the UK 4-digit SIC07 codes based on ONS conversion tables. In the last step, to mitigate double counting and measurement error, we apply the following rule: if two or more SIC03 codes with different degrees of tradability are mapped to one SIC07 code, we

label the SIC07 code using the highest tradable classification for the corresponding SIC03 codes. For example, if a SIC07 code maps to two tradable and a non-tradable SIC03 code, we classify that SIC07 code as tradable. If a SIC07 code maps to one tradable, one medium-tradable and one non-tradable SIC03 code, we also classify that SIC07 code as tradable.

Table A.18 above presents the resulting breakdown of firms by degree of tradability. Of the 93% of compliers that are in a manufacturing or services sector, we find that just over 23% are in manufacturing or highly tradable services, 46% are in medium-tradable and 23% in non-tradable services.

The remaining 7% of compliers either could not be classified (since IDBR data does not include a SIC07 code for these firms) or are classified in 'Other sectors' because they were excluded in Faggio & Overman (2014), such as firms in electricity and gas, transport, or telecommunications (roughly 62% of the 229 compliers in 'Other') or because they were not mapped in the correspondence process (e.g. some office and other administrative support activities, which account for 38% of the 229 compliers in 'Other'). The resulting 'Other' SIC07 would be considered non-tradable by some definitions.

## CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

1976	Julian Alves Bruno Serra Jason Greenberg Yaxin Guo Ravija Harjai John Van Reenen	Labour market power: New evidence on Non- Compete Agreements and the effects of M&A in the UK
1975	Virginia Minni	Global managers, local workers: Wage setting inside a multinational firm
1974	Jonathan Colmer Suvy Qin John Voorheis Reed Walker	The changing nature of pollution, income and environmental inequality in the United States
1973	Italo Colantone Gianmarco I.P. Ottaviano Piero Stanig	The social footprint of globalization: Towards the introduction of strategic industries in quantitative trade models
1972	Viet Nguyen-Tien Robert J.R. Elliott Eric Strobl Chengyu Zhang	Estimating the longevity of electric vehicles: What do 300 million MOT test results tell us?
1971	Michael Amior Jan Stuhler	Immigration, monopsony and the distribution of firm pay
1970	Jonathan Colmer David Lagakos Martin Shu	Is the electricity sector a weak link in development?
1969	Gaia Dossi Marta Morando	Political ideology and innovation
1968	Natalie Irmert Jan Bietenbeck Linn Mattisson Felix Weinhardt	Autonomous schools, achievement and segregation

1967	Stephen B. Billings Adam Soliman	The erosion of homeownership and minority wealth
1966	Pawel Bukowski Pawel Chrostek Filip Novokmet Marek Skawiński	Income inequality in the 21 <sup>st</sup> century Poland
1965	Benny Kleinman Ernest Liu Stephen J. Redding Motohiro Yogo	Neoclassical growth in an interdependent world
1964	Hanwei Huang Gianmarco I.P. Ottaviano	Rethinking revealed comparative advantage with micro and macro data
1963	Natalie Chen Dennis Novy Carlo Perroni Horng Chern Wong	Urban-biased structural change
1962	Robin Kaiji Gong Yao Amber Li Kalina Manova Stephen Teng Sun	Tickets to the global market: First US patent awards and Chinese firm exports
1961	Antonin Bergeaud Arthur Guillouzouic	Proximity of firms to scientific production
1960	Philippe Aghion Antonin Bergeaud Timothee Gigout Matthlieu Lequien Marc Melitz	Exporting ideas: Knowledge flows from expanding trade in goods
1959	Gabriel M. Ahlfeldt Nathaniel Baum-Snow Remi Jedwab	The skyscraper revolution: Global economic development and land savings

The Centre for Economic Performance Publications Unit Tel: +44 (0)20 7955 7673 Email <u>info@cep.lse.ac.uk</u> Website: <u>http://cep.lse.ac.uk</u> Twitter: @CEP\_LSE